# THE RATIONALITY PROBLEM FOR FIELDS OF INVARIANTS UNDER LINEAR ALGEBRAIC GROUPS (WITH SPECIAL REGARDS TO THE BRAUER GROUP)

## JEAN-LOUIS COLLIOT-THÉLÈNE\* AND JEAN-JACQUES SANSUC<sup>†</sup>

## INTRODUCTION

Let V be a vector space over an algebraically closed field k of characteristic zero. Let G be a reductive subgroup of GL(V). Assume that G acts almost freely on V, i.e. that for a general v in V, the stabilizer of G at v is trivial. Let k(V) denote the field of rational functions on V.

# **Problem.** Is the field of invariants $k(V)^G$ purely transcendental over k?

This is an old problem. In these notes, it will be referred to as the *purity* problem, or the rationality problem, indifferently. The question is open when G is a connected group. Even the case where G is the projective linear group  $PGL_n$  is not known, except for small values of n. However, when G is finite, the question has a negative answer, as was shown by D. Saltman in 1984. Saltman's paper was soon followed by a series of papers of F. Bogomolov, then by further papers of D. Saltman. Between 1986 and 1988 we ran seminars on their work, and in July 1988 one of us lectured on this topic at the IX Escuela Latinoamericana de Matemáticas, held in Santiago de Chile, and a set of notes was distributed. Over the years, versions of these notes were circulated and used as a complement to the original work of Saltman and Bogomolov. We are grateful to the organizers of the 2004 International Conference in Mumbai for giving us the opportunity to publish a revised version of our text. We have not tried to update the notes systematically, but we have added references to work done since 1988. The rationality problem may also be raised over a field k which is not algebraically closed. The interested reader is referred to [36, 37, 92, 117, 146, 147, 153].

Here is the list of sections. Sections 1 to 4 are devoted to general definitions and results, whereas Sections 5 to 9 concentrate on the computation of the unramified Brauer group.

# Contents.

- 1. Rationality.
- 2. Quotients.
- 3. General techniques.
- 4. Examples.
- 5. The unramified Brauer group.

Date: September 19, 2005.

6. A general formula.

- 7. Linear action of a finite group.
- 8. Multiplicative action of a finite group.
- 9. Homogeneous spaces.

Section 1 introduce notions close to "rationality": unirationality, stable rationality, retract rationality. Definitions and properties are given in terms of function fields — the algebraic side — and in terms of algebraic varieties — the geometric side.

Section 2 deals with fields of invariants  $K^G$  under a linear algebraic group G acting on a function field K. It also deals with "geometric models" of such an action, i.e. a G-action on an integral k-variety X such that K coincides with the function field k(X) of X and  $K^G$  coincides with the function field k(X/G) of some "good" quotient X/G. The most relevant concept for these notes is that of an "almost free" action — sometimes also called "generically free" action.

Section 3 discusses two basic techniques: a "slice" technique and a lemma which has since then gone into the literature as the "no-name lemma". The "no-name lemma" says that for "almost-free" linear G-actions the (stable) purity question depends only on G. The "slice method" enables one to see the field of invariants for an action of a given group G as the field of invariants for another action of a smaller group H.

Section 4 considers linear actions of some particular groups for which the (stable) rationality problem has a positive answer: some finite groups ( $\mathfrak{S}_n, \mathfrak{A}_5$ ), solvable groups, the so-called "special groups" ( $\mathrm{GL}_n, \mathrm{SL}_n, \mathrm{Sp}_{2n}$ ), the orthogonal groups  $\mathrm{O}_n, \mathrm{SO}_n$ . For the spinor groups  $\mathrm{Spin}_n$  and for  $\mathrm{PGL}_n$ , there are results in low dimension.

The unramified Brauer group of a function field K/k is defined in Section 5 and some of its basic properties are given. A key property is that if K is a purely transcendental extension of k, then the unramified Brauer group of K (over k) is trivial.

In Section 6, a quite general formula of Bogomolov for the unramified Brauer group of the field of invariants of an (almost free) action of a group G is given, first for G finite then more generally for G reductive. The finite bicyclic subgroups of G play a key rôle in this computation.

In Section 7, this formula is applied to the case of a linear action of a finite group. It is further specialized to the case of nilpotent groups of class 2, where it yields concrete examples of fields of linear invariants which are not rational, among which one finds Saltman's original example.

Section 8 discusses (twisted) multiplicative invariants of a finite group G, after Saltman and others. In favourable circumstances, one may view the associated field of G-invariants as the field of invariants of a linear action of a finite group G' which is an extension of G. This is used to produce other types of fields of linear invariants which are not rational.

Finally, Section 9 gives in some detail Bogomolov's proof of the vanishing of the unramified Brauer group of the function field of a quotient G/H, where H is a connected subgroup of a simply connected group G. One first proves

the vanishing of the unramified Brauer group of the field  $k(V)^H$  for an almost free linear representation V of H.

Precise references to the papers of Saltman and Bogomolov are given in the text. Some of our proofs differ from the original ones.

Known examples of non-rationality. For the action of a finite group G on a purely transcendental extension K of  $k = \mathbb{C}$ , non-rational fields of invariants  $K^G$  of the following types have been exhibited :

- (i) Linear action of a nilpotent group of order  $p^7$  and class 2.
- (ii) Multiplicative actions of  $G = (\mathbb{Z}/p)^3$ .
- (iii) Twisted multiplicative actions of  $G = (\mathbb{Z}/p)^2$ .
- (iv) For an arbitrary action,  $G = \mathbb{Z}/2$  (Clemens-Griffiths [31], Artin-Mumford [3]).

Over a non-algebraically closed field k, there exist examples of non-rational fields of invariants  $K^G$  for an almost free linear action of the following kinds:

- (v)  $k = \mathbb{Q}, G = \mathbb{Z}/47$  (Swan [146], Voskresenskii [153]).
- (vi)  $k = \mathbb{Q}, G = \mathbb{Z}/8$  (Saltman [117], Voskresenskii [153]).
- (vii)  $k = \mathbb{Q}, G \text{ a } k$ -torus [36].

(viii) G a simply connected semisimple group (Merkurjev [92])

But for almost free linear actions of connected linear algebraic groups over the complex field, the rationality question for fields of invariants is open.

# 1. RATIONALITY

Let k be a field. Let  $\mathbb{A}^n$ , resp.  $\mathbb{P}^n$ , denote the *n*-dimensional affine space, resp. projective space, over k. The function field of an integral k-variety X is denoted k(X). The set of rational k-points of X is denoted X(k).

One says that two integral k-varieties X and Y are k-birationally equivalent if the following equivalent conditions are satisfied:

- (i) The function fields k(X) and k(Y) are isomorphic (over k).
- (ii) There exist non-empty Zariski open sets  $U \subset X$  and  $V \subset Y$  which are isomorphic over k.

One says that two integral k-varieties X and Y are stably k-birationally equivalent if the following equivalent conditions are satisfied:

- (i) For suitable integers r, s and independent variables  $\{x_i\}_{i=1}^r$  and  $\{y_j\}_{j=1}^s$ , the fields  $k(X)(x_1, \ldots, x_r)$  and  $k(Y)(y_1, \ldots, y_s)$  are k-isomorphic (one then says that the fields k(X) and k(Y) are stably equivalent).
- (ii) For suitable integers r, s, the k-varieties  $X \times_k \mathbb{A}^r$  and  $Y \times_k \mathbb{A}^s$  are k-birationally equivalent.

A k-variety X is said to be

- k-rational if it is integral and it is k-birational to an affine space (one then says that k(X) is pure over k),
- stably k-rational if there exists an affine space  $\mathbb{A}^n$  over k such that  $X \times_k \mathbb{A}^n$  is k-rational (one then says that k(X) is stably pure over k),

- a direct factor of a k-rational variety if there exists an integral k-variety Y such that  $X \times_k Y$  is a k-rational variety.
- *k*-unirational if it is integral and it satisfies one of the equivalent properties in the following lemma.

**Lemma 1.1.** Let X be an integral k-variety. The following conditions are equivalent:

- (i) The function field k(X) of X is a k-subfield of a pure extension K of k.
- (ii) There exists a dominating k-morphism from a k-rational variety Y to X.
- (iii) (Under the additional assumption that k is infinite) there exists a dominating k-morphism from a k-rational variety Y to X, with  $\dim Y = \dim X$ .

Proof. We only need to show that (ii) implies (iii). Let  $f: W \to X$  be a dominating k-morphism, where  $W \subset \mathbb{A}^n$  is an open set of affine space over k. By linear subspace of W we shall mean the non-empty trace on W of a linear subspace of  $\mathbb{A}^n$ . Let us consider the closed, geometrically integral subvarieties  $Y \subset X$  over k with the following property: there exists a linear subspace  $V \subset W$  with dim  $V = \dim Y$  such that the k-morphism f restricts to a dominant k-morphism from V to Y. Any k-point in f(W(k)) is of this type. Let Y be such a variety. Assume  $Y \neq X$ . Since k is infinite and f is dominant, there exists a k-point  $P \in W(k)$  such that  $M = f(P) \in X(k)$  does not lie on Y. Since f is defined at the generic point of V, it is also defined at the generic point of the linear span  $L \subset W$  of V and P. Moreover, the closure  $Y_1 \subset X$  of the image of L under f contains Y and M, hence is of dimension strictly bigger than Y. Iterating this procedure, we find that X is covered by a k-linear space of dimension dim X.

Finally, following Saltman, one says that a k-variety X is

• retract rational (over k) if it satisfies one of the equivalent conditions in the following proposition (one then says that k(X) is retract rational over k).

**Proposition 1.2** (Saltman [118, Theorem 3.8]). Let k be a field and X be an integral k-variety. The following conditions are equivalent:

- (i) There exists a non-empty open set U of X such that the identity morphism of U factorizes through a Zariski open set Y of an affine space over k, i.e. there are maps U → Y → U whose composite is identity on U.
- (ii) There exists a non-empty open set V of X such that for any local kalgebra A with residue field  $\kappa$ , the natural map  $V(A) \to V(\kappa)$  is onto.

*Proof.* The surjectivity of the natural map  $V(A) \to V(\kappa)$  means that any map

Spec 
$$\kappa \to V$$

extends to  $\operatorname{Spec} A$ :

$$\operatorname{Spec} \kappa \to \operatorname{Spec} A \to V.$$

If U is a Zariski open set of V, it is stable by generisation, hence the surjectivity of  $V(A) \to V(\kappa)$  implies that of  $U(A) \to U(\kappa)$ .

That (i) implies (ii) is now clear. For an affine space  $Y = \mathbb{A}^n$  the map  $Y(A) \to Y(\kappa)$  is the natural reduction  $A^n \to \kappa^n$  and it is surjective. Thus this is also true for the open set Y of affine space, hence also for U.

In order to prove the converse assertion, we may assume that X = V and that X is affine, say  $X = \operatorname{Spec} B$  with  $B = R/\mathfrak{p}$ , with  $\mathfrak{p}$  a prime ideal of the k-algebra  $R = k[x_1, \ldots, x_n]$  of polynomials in n variables  $x_1, \ldots, x_n$ . Let K = k(X) be the function field of X. This is also the residue class field of the local ring  $A = R_\mathfrak{p}$ . The generic point  $\eta$  of X defines a point in X(K). The assumption implies that this point comes from an element of  $X(R_\mathfrak{p})$ , i.e. we have maps

$$\eta \to \operatorname{Spec} R_{\mathfrak{p}} \to X$$

whose composite is the natural inclusion of  $\eta$  into X. The map  $\operatorname{Spec} R_{\mathfrak{p}} \to X$  factorizes through some  $\operatorname{Spec} R_f$ , where  $R_f = R[\frac{1}{f}]$ , i.e. we have an open subset  $V \subset \mathbb{A}^n$  and maps

$$\eta \to V \xrightarrow{\pi} X$$

whose composite is the natural inclusion of  $\eta$  into X. Then this situation extends to an open set U of X, i.e. there exist a dense open set  $U \subset X$  and maps

$$U \to V \xrightarrow{\pi} X$$

whose composite is the natural inclusion of U into X. Setting  $Y = \pi^{-1}(U)$  gives rise to the announced factorization

$$U \to Y \to U.$$

Remark 1.3. Saltman's motivation for introducing the concept of retract rationality was to try to understand the relation between rationality and approximation properties (in a more arithmetical context). The criterion above, though nice, seems a priori to be of little value: for instance it seems impossible to prove that a smooth affine conic satisfies the lifting property (ii) without a priori proving that it is a rational curve! Nevertheless, in the particular case of an almost free (see §2) linear action of the projective linear group  $PGL_p$ with p prime, Saltman used (ii) to prove that the quotient is retract rational [118, Corollary 5.3] — but a direct proof that this variety satisfies (i) could later be given [37, Corollary 9.13].

**Proposition 1.4.** Let X be an integral k-variety. In the list of properties:

- (i) X is k-rational,
- (ii) X is stably k-rational,
- (iii) X is a direct factor of a k-rational variety,
- (iv) X is retract rational over k,
- (v) X is k-unirational,

each property implies the following one.

Proof. Only the penultimate implication (iii)  $\implies$  (iv) requires a proof. Let Y be an integral k-variety such that  $X \times_k Y$  is rational. Let  $U \subset X \times_k Y$  be a non-empty open set which is isomorphic to an open set of affine space. Let  $(x_0, y_0) \in U(k)$ . Let  $X_1 \subset X$  be the non-empty open set such that  $X_1 \times \{y_0\} = U \cap (X \times \{y_0\})$ . The open set  $U_1 = U \cap (X_1 \times Y)$  is still isomorphic to an open set of affine space. Now the composite map  $X_1 \to U_1 \to X_1$ , where the first map is given by  $x \mapsto (x, y_0)$  and the second map is induced by projection onto X, satisfies the requirements of Proposition 1.2 (i).

*Remarks.* Over an algebraically closed field of characteristic zero, examples are known of:

- (a) unirational varieties which are not rational (Artin-Mumford [3], Iskovskikh-Manin [57], Clemens-Griffiths [31]);
- (b) stably rational varieties which are not rational [6].

The Artin-Mumford method for proving non-rationality uses the Brauer group. This method will be discussed and applied in §§5–9 of these notes. The method applies to varieties of arbitrary dimension. The associated invariant is insensitive to replacement of the variety X by  $X \times_k \mathbb{A}^n$ . Such is not the case for the two other methods mentioned.

The Clemens-Griffiths method is quite specific to threefolds. That method, in a variant due to Mumford, was used to produce the examples in (b).

The Iskovskikh-Manin method was originally developped to prove the nonrationality of the general quartic threefold. It has witnessed a strong development: birational rigidity and birational super-rigidity, work of Iskovskikh [56], Pukhlikov [112, 113], de Fernex-Ein-Mustață [38].

One should here also mention two further techniques for disproving rationality. One is a natural generalization of the Artin-Mumford technique: it uses higher dimensional unramified cohomology (with torsion coefficients). For this, see [32,35,92,98–100,128–130]. The other one, due to Kollár [75,76], uses reduction to positive characteristic.

### 2. Quotients

In this section k denotes a field of characteristic zero and k an algebraic closure of k. We denote by  $\mathfrak{g}$  the Galois group of  $\bar{k}$  over k. Given a k-variety X we denote  $\bar{X} = X \times_k \bar{k}$ . If X is integral, resp. geometrically integral, we denote by k(X), resp.  $\bar{k}(X)$ , the function field of X, resp.  $\bar{X}$ .

Let G be a linear algebraic group over k and let X be a geometrically integral k-variety with a G-action. Our interest will be in the field  $k(X)^G$ , which by definition is the fixed field of  $\mathfrak{g}$  acting on the field of invariants  $\bar{k}(X)^{G(\bar{k})}$ .

2.1. Elementary properties. The ring of regular functions on a k-variety X is denoted k[X].

**Lemma 2.1.** Let A be a domain, K its field of fractions, G a finite group acting on A. Then the field  $K^G$  is the field of fractions of  $A^G$ .

Proof. Let  $f \in K^G$ . Write f = a/b with  $a, b \in A$ . Let  $e \in G$  be the identity element. Since A is commutative,  $\beta := N(b) = \prod_{\sigma \in G} {}^{\sigma}b \in A^G$ . Moreover,  $\beta = bc$  with  $c := \prod_{\sigma \in G, \sigma \neq e} {}^{\sigma}b \in A$ . We have  $f = \alpha/\beta$  with  $\alpha := ac \in A$ . Since f and  $\beta$  are G-invariants,  $\alpha = f\beta$  is also G-invariant. Finally,  $f = \alpha/\beta$  with  $\alpha, \beta \in A^G$ .

**Lemma 2.2.** Let  $k = \overline{k}$ . Let A be an integral k-algebra of finite type, equipped with an (algebraic) action of a linear algebraic k-group G. Let K be the field of fractions of A. Assume that A is a UFD and that the identity component of G has no nontrivial character. Then  $K^G$  is the field of fractions of  $A^G$ .

See [95, Theorem 4.1; 108, Lemma 3.2 and Theorem 3.3].

*Proof.* (i) Let us first assume that G is connected. For each prime ideal of height one  $\mathfrak{p}$  of A, fix a generator  $f_{\mathfrak{p}} \in A$ . Let  $f \in K^G$ . Write

$$f = u f_1^{r_1} \dots f_s^{r_s}$$

with  $u \in A^*$ , the  $f_i = f_{\mathfrak{p}_i}$  among the generators chosen above and the  $r_i \in \mathbb{Z}$ ,  $r_i \neq 0$ . For any  $\sigma \in G(k)$ , the identity  ${}^{\sigma}f = f$  implies

$$f = u f_1^{r_1} \dots f_s^{r_s} = {}^{\sigma} u \cdot {}^{\sigma} f_1^{r_1} \dots {}^{\sigma} f_s^{r_s}.$$

Then, by unique factorization, for each  $i = 1, \ldots, s$  there exists a  $j = \tau_{\sigma}(i)$ such that  ${}^{\sigma}f_i = \varepsilon_i(\sigma)f_j$ , with  $\varepsilon_i(\sigma) \in A^*$ . Since G is connected, the induced homomorphism  $\tau: G \to \mathfrak{S}_s$  is trivial. For each i, we thus have  ${}^{\sigma}f_i = \varepsilon_i(\sigma)f_i$ . Let  $\varepsilon_i: G(k) \to A^*$  be the induced homomorphism, and  $\pi: A^* \to A^*/k^*$ . Since G is connected and  $A^*/k^*$  is an abelian group of finite type (see [36]),  $\pi \circ \varepsilon_i$  is trivial. Hence  $\varepsilon_i$  is a homomorphism  $\varepsilon_i: G(k) \to k^*$ . All data being algebraic, it is induced by a character  $G \to \mathbb{G}_m$ , hence trivial by assumption. Thus each  $f_i$  is G-invariant. This then implies  $u \in A^G$ , and f lies in the fraction field of  $A^G$ .

(ii) Let G be arbitrary. Let  $G^{\circ}$  be the identity component of G. Let  $f \in K^G$ . By (i), f = a/b with  $a, b \in A^{G^{\circ}}$ . The same arguments as in the proof of Lemma 2.1 for the finite group  $G/G^{\circ}$  show that  $f = \alpha/\beta$  with  $\alpha \in A$  and  $\beta \in A^G$ , hence also  $\alpha \in A^G$ .

*Remarks.* (1) The above arguments also shows that  $A^G$  is a UFD.

(2) The assumption that the identity component of G has no nontrivial character is a necessary one, as the example of the diagonal action of  $\mathbb{G}_m$  on  $\mathbb{A}_k^2$  shows.

**Proposition 2.3.** Let  $k = \overline{k}$ . Let G be a linear algebraic group and X a factorial affine variety with a G-action.

- (i) If  $k[X]^* = k^*$ , there exists a G-invariant affine open set  $U \subset X$  such that  $k(X)^G$  is the field of fractions of  $k[U]^G$ .
- (ii) If G is finite, or if G has no nontrivial character  $G \xrightarrow{\chi} \mathbb{G}_m$ , then  $k(X)^G$  is the field of fractions of  $k[X]^G$ .

See also [108, Theorem 3.3].

Proof. (i) Let  $k(X)^G = f(f_1, \ldots, f_s)$ . Let Z be the union of the supports of the divisors of the functions  $f_i$ . Since X is factorial affine, there exists  $g \in k[X]$  whose divisor is Z. The complement U of Z is therefore an affine open set with k[U] = k[X][1/g]. The divisor Z being G-invariant, U is G-invariant. Since  $f_1, \ldots, f_s$  belong to  $k[U]^G$ , this implies  $k(X)^G$  is the field of fractions of  $k[U]^G$ .

Assertion (ii) is just a rephrasing of Lemmas 2.1 and 2.2.

In order to describe a field of invariants  $k(X)^G$ , it will often be convenient to realize it as the function field of a "quotient" variety Y of X by G.

In the literature, several integral varieties Y equipped with a morphism  $p: X \to Y$  such that p(gx) = p(x) for  $g \in G$  and  $x \in X$  go under the name of quotient varieties. However, our main interest is in such quotients with the additional property  $k(Y) = k(X)^G$ .

**Proposition 2.4.** Let  $\pi: X \to Y$  be a dominant morphism of geometrically integral k-varieties. Let the algebraic k-group G act on X. Assume that for  $x, y \in X(\bar{k})$ , the following two conditions are equivalent:

(i)  $\pi(x) = \pi(y);$ 

(ii) there exists  $g \in G(\bar{k})$  such that gx = y.

Then k(Y) may be identified with the field  $k(X)^G$  of invariants:

 $k(Y) = k(X)^G.$ 

*Proof.* For the proof, we may assume  $k = \bar{k}$ . In this case the proposition is an immediate consequence of the following elementary proposition — whose proof uses the characteristic zero hypothesis.

**Proposition 2.5** (see [83, AI.3.7 Satz 2]). Let  $k = \bar{k}$ . Let  $\phi: X \to Y$  be a dominant morphism of integral k-varieties. If  $f \in k(X)$  is a rational function which is constant on the fibres of  $\phi$ , then f may be identified with a rational function on Y.

### 2.2. Algebraic quotients, geometric quotients, torsors.

**Definition 2.6.** If G is a reductive k-group and  $X = \operatorname{Spec} A$  is an affine k-variety with a G-action, the algebraic quotient  $X/\!\!/G$  of X by G is the affine scheme  $\operatorname{Spec} A^G$ . This actually is a k-variety, since G reductive implies (see [83, II.3.2; 94, Chap. 1, §2]) that  $A^G$  is a finitely generated k-algebra.

**Definition 2.7.** Let X be an algebraic k-variety equipped with an action of an algebraic k-group G. A geometric quotient of X by G is a k-variety Y equipped with a k-morphism  $\phi: X \to Y$  such that:

- (i)  $\phi$  is open, constant on *G*-orbits and induces a bijection of  $X(\bar{k})/G(\bar{k})$  with  $Y(\bar{k})$ .
- (ii) For any open subset  $V \subset Y$ , the natural morphism  $k[V] \to k[\phi^{-1}(V)]^G$  is an isomorphism.

If such a variety Y exists, it is unique up to unique isomorphism. It will then be denoted Y = X/G. By a classical theorem of Rosenlicht [115], given X an integral algebraic variety equipped with an action of a linear algebraic group G, there exists a nonempty G-stable open set  $U \subset X$  admitting a geometric quotient  $U \to U/G$ .

*Remark.* A geometric quotient  $(Y, \phi)$  of X by G has the following properties:

- (a) The map  $\phi: X(\bar{k}) \to Y(\bar{k})$  is onto, all  $G(\bar{k})$ -orbits on  $X(\bar{k})$  are closed and they coincide with the fibres of  $\phi$  over  $\bar{k}$ -points.
- (b) Proposition 2.4 implies  $k(Y) = k(X)^G$ .
- (c) The map  $\phi: X \to Y$  has the obvious universal mapping property [23, p. 172]. This accounts for the unicity statement above.
- (d) When X is affine and G is reductive, the algebraic quotient  $X/\!\!/G$  need not be a geometric quotient (consider the diagonal action of  $\mathbb{G}_m$  on  $\mathbb{A}^2$ ).

**Proposition 2.8.** Let k = k. Let X be a normal algebraic k-variety equipped with an action of an algebraic k-group G. Then a k-morphism  $\phi: X \to Y$  is a geometric quotient if and only if

- (1)  $\phi$  is constant on G-orbits.
- (2)  $\phi$  induces a bijection  $X(k)/G(k) \xrightarrow{\cong} Y(k)$ .
- (3) Y is normal.

*Proof.* The proof uses the char k = 0 hypothesis and Zariski's main theorem, see [94, Prop. 0.2, p. 7].

**Definition 2.9.** Let G be a linear algebraic k-group. A k-variety X equipped with a G-action and a faithfully flat k-morphism  $X \to Y$  is a G-torsor principal homogeneous space under G — over Y if the map  $(g, x) \mapsto (gx, x)$ induces an isomorphism  $G \times_k X \cong X \times_Y X$ .

Remark 2.10. If X is affine, then Y is affine and  $Y = X/\!\!/G$ .

Remark 2.11. If X is a G-torsor over Y the map  $X \to Y$  makes Y into a geometric quotient X/G. Hence  $k(Y) = k(X)^G$ .

Remark 2.12. Let G be a k-algebraic group and H a subgroup. The geometric quotient G/H exists, and  $G \to G/H$  makes G into an H-torsor over G/H. Hence  $k(G/H) = k(G)^H$ .

There is a vast literature on the various possible notions of quotient. We refer the reader to [13, 14, 40, 83, 108].

## 2.3. Almost free actions.

**Definition 2.13.** An action of an algebraic k-group G on a geometrically integral k-variety X will be called an *almost free action* if for all  $x \in U(\bar{k})$  in a non-empty open set U of X the stabilizer  $G_x \subset G(\bar{k})$  is trivial.

**Theorem 2.14** (Luna [87]). Let X be an affine k-variety and let the reductive k-group G act on X. If each stabilizer  $G_x$ ,  $x \in X(\bar{k})$ , is trivial, then X is a G-torsor over  $X/\!\!/G$ . Hence, X/G exists, and  $X/G = X/\!\!/G$ .

*Remark.* The hypothesis "X affine" cannot be ignored (see [94, p. 11]).

**Theorem 2.15** (Luna [87]). Let k = k. Let the reductive group G act upon the affine variety X. Let  $\pi: X \to X/\!\!/G$  be the natural projection. If there exists a point  $x \in X(k)$  with trivial stabilizer and closed orbit, then there exists an affine open set  $V \subset X/\!\!/G$  which contains  $\pi(x)$  and is such that the projection  $\pi^{-1}(V) \to V$  makes  $\pi^{-1}(V)$  into a G-torsor over V.

*Proof.* This is an immediate consequence of Luna's "slice theorem" [87].  $\Box$ 

**Corollary 2.16.** Let  $k = \overline{k}$ . Let the reductive group G act upon the variety X. The following conditions are equivalent:

- (i) There exist a G-invariant non-empty open set  $U \subset X$  and a variety Y with a morphism  $U \to Y$  such that U is a G-torsor over Y.
- (ii) There exists a G-invariant non-empty affine open set  $U \subset X$  such that the projection map  $U \to U/\!\!/ G$  makes U into a G-torsor over  $U/\!\!/ G$ .
- (iii) There exist an affine G-invariant open set U and a point  $x \in U(k)$  such that the stabilizer  $G_x$  is trivial and the orbit of x in U is closed.

*Proof.* To show that (i)  $\Longrightarrow$  (ii), choose an affine non-empty open set  $V \subset Y$  and take its inverse image  $\pi^{-1}(V)$  under  $\pi: U \to Y$ . Conversely, (ii)  $\Longrightarrow$  (i) is obvious. It is also obvious that (ii)  $\Longrightarrow$  (iii). Finally, (iii)  $\Longrightarrow$  (ii) is a consequence of Theorem 2.15.

**Theorem 2.17** (Popov [103], Luna-Vust [89], see [94, appendix p. 154]). Let  $k = \bar{k}$ . Let the semisimple group G act on the factorial affine variety X. The following conditions are equivalent:

- (i) There exists a G-invariant non-empty open set U ⊂ X such that for each x ∈ U(k), the stabilizer G<sub>x</sub> is reductive.
- (ii) There exists a G-invariant non-empty open set  $V \subset X$  such that for each  $x \in V(k)$ , the orbit  $G \cdot x$  of x is closed in X.

As the diagonal action of the multiplicative group  $\mathbb{G}_m$  on affine space reveals, Theorem 2.17 does not extend to reductive groups. We thank M. van den Bergh for showing us how to prove the next theorem for arbitrary reductive groups.

**Theorem 2.18.** Let k = k. Let X be a factorial affine variety, and let the reductive group G act on X. Assume that the open set consisting of points with finite stabilizer is non-empty.

Then there exist non-empty G-invariant open sets  $U_1 \subset U \subset X$  such that U is affine and G-orbits of points of  $U_1$  are closed in U.

Proof. Let  $G^{\circ} \subset G$  be the connected component of identity in G. Assume we have found suitable  $U_1 \subset U$  for  $G^{\circ}$ . Let V be the intersection of all  $\sigma U$ , for  $\sigma$  running through a (finite) system of representatives in G for  $G/G^{\circ}$ . The open set V is affine and G-stable. Let  $V_1 = U_1 \cap V$ . The G-orbits of points of  $V_1$  are finite unions of orbits of  $G^{\circ}$ , hence are closed in U, hence also in V. We may therefore assume that G is connected.

Let  $T = Z(G)^{\circ}$  be the torus which is the connected centre of G, and let  $G_0 = [G, G]$  be the derived subgroup of G. It is a connected semisimple group.

Then G is the almost direct product of T and  $G_0$ , i.e. it is the quotient of  $T \times G_0$ by the finite group  $= T \cap G_0$ . Let  $f: X \to X/\!\!/G_0$  be the algebraic quotient of X by  $G_0$ . The torus T acts on  $X/\!\!/G_0$  and the map f is T-equivariant. Any point of X with finite stabilizer in G projects down to a point of  $X/\!\!/G_0$ with finite stabilizer in T. Hence there exists a non-empty T-stable open set  $W_0 \subset X/\!\!/G_0$  consisting of points with finite stabilizer. All orbits of T on  $W_0$ are closed (if an orbit of T on  $W_0$  were not closed, its closure would contain a smaller dimensional orbit, hence points with positive dimensional stabilizer).

Since T is a torus, a theorem of Sumihiro [145] asserts that there exists a non-empty affine T-invariant open set  $U_0 \subset W_0$ . Clearly, T-orbits are closed in  $U_0$ . Let  $U = f^{-1}(U_0)$ . Since f is affine, this is a G-invariant affine open set of X. Since X is factorial, so is U. By assumption, X and hence U contains a non-empty G-stable open set consisting of points with finite stabilizer. Thus according to Theorem 2.17 there exists a non-empty  $G_0$ -stable open set  $U_1$  of U consisting of points with closed  $G_0$ -orbit in U and finite stabilizer. For any point  $x \in U_1(k)$ ,  $G_0x$  is of maximal dimension in  $f^{-1}(f(x))$ . Since it is closed, it coincides with  $f^{-1}(f(x))$ . Now

$$Gx = TG_0 x = T(f^{-1}(f(x))) = f^{-1}(Tf(x)))$$

is the inverse image of the closed set  $Tf(x) \subset U_1$ , hence is closed.

**Corollary 2.19.** Let X be an affine k-variety. Assume that  $\overline{X}$  is factorial. Let the reductive k-group G act almost freely on X. Then there exists a Ginvariant non-empty affine open set  $U \subset X$  such that the projection map  $U \to U/\!\!/G$  makes U into a G-torsor over  $U/\!\!/G$ .

*Proof.* To prove this, we may assume  $k = \overline{k}$ . According to Theorem 2.18, there exist non-empty *G*-invariant open sets  $U_1 \subset U \subset X$  such that *U* is affine and  $U_1$  consists of points *x* with trivial stabilizer and closed orbit in *U*. It only remains to apply Corollary 2.16, (iii)  $\Longrightarrow$  (i).

# 3. General techniques

In this section, k denotes a field of characteristic zero and k an algebraic closure of k.

Several methods have been used to prove the rationality of some quotient spaces, in particular by Katsylo. A good survey of these methods is given by Dolgachev [41], to whom we refer for many examples from the theory of moduli spaces. In this section, these general methods will be reviewed; an attempt will be made to give complete proofs of some of the basic lemmas.

3.1. The slice method. The following very useful fact, which is known under the name "slice method" goes back at least to C. S. Seshadri [138]. See M. Nagata [96, Lemma 1 ("Lemma of Seshadri"), p. 37] and also V. L. Popov [106]. The following version of Seshadri's lemma comes from [46].

**Theorem 3.1** (Slice lemma). Let G be an algebraic group over k and X a geometrically integral k-variety with a G-action. Let  $Y \subset X$  be a closed

geometrically integral sub-k-variety, and let  $H \subset G$  be the normalizer of Y. Assume:

- (i) The closure of  $G \cdot Y$  is the whole of X.
- (ii) There exists a non-empty H-stable open subset  $Y_0$  of Y such that if  $g \in G(\bar{k})$  and  $y \in Y_0(\bar{k})$  satisfy  $g \cdot y \in Y_0(\bar{k})$ , then there exists  $h \in H(\bar{k})$  such that  $h \cdot y = g \cdot y$ .

Then there is a natural k-isomorphism of fields  $k(X)^G \cong k(Y)^H$ .

Proof. We immediately reduce to the case  $k = \bar{k}$ . Let  $f \in k(X)^G$ ,  $f \neq 0$ . Since  $G \cdot Y$  is dense in X, the subvariety Y is not contained in the support of div f. Hence  $f|_Y \in k(Y)$  and the restriction defines an injection  $k(X)^G \subset k(Y)^H$ . Conversely, let  $f \in k(Y)^H = k(Y_0)^H$ . Let  $\pi \colon G \times Y_0 \to X$  be the natural

Conversely, let  $f \in k(Y)^H = k(Y_0)^H$ . Let  $\pi: G \times Y_0 \to X$  be the natural morphism. The rational function  $\phi$  defined on  $G \times Y_0$  by  $\phi(g, y) = f(y)$  is constant on the fibres of  $\pi$  (use assumption (ii) and the *H*-invariance of *f*). Since  $\pi$  is a dominant morphism of integral varieties,  $\phi$  may be identified see Proposition 2.5 — with a rational function  $\tilde{f}$  on *X*, and  $\tilde{f}|_Y = f$ . Hence  $k(X)^G = k(Y)^H$ .

*Remark* 3.2. We refer to [107] for more historical remarks regarding this theorem, which may be consider as a birational analogue of the Chevalley restriction theorem [88].

For other reduction techniques, see [81, 82].

3.2. The no-name lemma ([20, 41]). In quite a few circumstances, the method to be described below enables one to deduce stable rationality and in some cases rationality of some fields of invariants once it is known for other fields. This method has been independently discovered by several people, hence the denomination "no-name lemma" given by Dolgachev [41] to one of the corollaries below.

We first give statements and proofs for finite groups actions. Then we give an independent proof for the general case of reductive group actions.

**Theorem 3.3** (Speiser's lemma). Let G be a finite group. Let V and W be two faithful finite-dimensional linear representations of G over k. Then the fields  $k(V)^G$  and  $k(W)^G$  are stably equivalent.

This means that there exist independent variables  $x_1, \ldots, x_r$  and  $y_1, \ldots, y_s$  such that

$$k(V)^G(x_1,\ldots,x_r) \cong k(W)^G(y_1,\ldots,y_s).$$

**Corollary 3.4.** Let V be a faithful finite-dimensional linear representation of a finite group G over k. The stable purity of the field  $k(V)^G$  depends only on G.

We have the well-known lemma:

**Lemma 3.5.** Let K/k be a finite Galois extension. Let G be its Galois group. Then for any finite dimensional K-linear space E and any semi-linear action of G on E with respect to the extension K/k we have

$$E = E^G \otimes_k K.$$

In particular,  $K(E)^G/k = k(E^G)/k$  is pure.

*Proof.* Let us prove the last assertion. If  $E = E^G \otimes_k K$ , we are reduced to the case  $E = K^n$  with the natural action of G. Hence  $K(E) = K(t_1, \ldots, t_n)$  with trivial action of G on the  $t_i$ 's. So  $K(E)^G/k = k(t_1, \ldots, t_n)/k$ .

For the first assertion, let  $\{\omega_i\}_{i=1,\ldots,d}$  be a linear basis of K/k and  $G = \{\sigma_i\}_{i=1,\ldots,d}$  with  $\sigma_1 = \mathrm{id}_K$ . Let  $v \in E$ . Consider, for every  $i = 1, \ldots, d$ ,

$$w_i := \sum_j \sigma_j(\omega_i v) \in E^G.$$

The group G acting semi-linearly on E with respect to K/k, we have

$$w_i = \sum_j \sigma_j(\omega_i)\sigma_j(v).$$

The matrix  $(\sigma_j(\omega_i))$  being invertible, one can express  $v = \sigma_1(v)$  as a linear combination of the  $w_j$ 's with coefficients in K, i.e.  $v \in E^G \otimes_k K$ . Thus the natural map

$$E^G \otimes_k K \to E$$

is onto. It remains to show that this map is injective. Let  $\sum_i \lambda_i v_i = 0$  be a non-trivial K-linear relation between elements  $v_i \in E^G$ . There exist  $i_0$  and  $\alpha \in K$  with  $\operatorname{Tr}_{K/k}(\alpha \lambda_{i_0}) \neq 0$ . Then  $\sum_i \operatorname{Tr}_{K/k}(\alpha \lambda_i) v_i = 0$  is a non-trivial k-linear relation between the  $v_i$ 's.

Proof of Theorem 3.3. By hypothesis, the linear representation of G on V is faithful. As a consequence the extension  $k(V)/k(V)^G$  is Galois of group G. Then we can apply Lemma 3.5 to the following data: the extension  $k(V)/k(V)^G$  and the k(V)-linear space  $E = W_{k(V)}$ . Lemma 3.5 ensures the purity of the extension

$$k(V \oplus W)^G / k(V)^G = \left(k(V)(W)\right)^G / k(V)^G.$$

By exchange of V and W we also obtain the purity of  $k(V \oplus W)^G/k(W)^G$ . Hence  $k(V \oplus W)^G$  is a common pure extension of  $k(V)^G$  and of  $k(W)^G$ .  $\Box$ 

We now give statements and proofs for the general case.

**Theorem 3.6.** Let G be a reductive group over k. Let Y = Spec A be an affine k-variety with a G-action. Assume that all geometric stabilizers at closed points of X are trivial. Let  $X \to Y$  be a vector bundle over Y equipped with an equivariant G-action. Then:

(a)  $X/\!\!/G$  is a vector bundle over  $Y/\!\!/G$ .

(b) If Y and hence X are integral, the field  $k(X)^G$  is pure over  $k(Y)^G$ .

*Remark* 3.7. In the paper [20] by Bogomolov and Katsylo, no condition is imposed upon the linear group G. Also, the base variety Y is not assumed to be affine. Although this last extension is perfectly legitimate, one should be aware that there are actions of SL(2) on non-affine varieties such that:

- (i) All stabilizers are trivial, i.e., the action is set-theoretically free.
- (ii) There is a geometric quotient.

(iii) The action is not scheme-theoretically free (Mumford, [94, p. 11]), i.e., the morphism  $G \times X \to X \times X$  given by  $(g, x) \mapsto (gx, x)$  is not a closed immersion.

For a proof of part (b) for arbitrary G and Y, see [27, §4.3].

*Proof of Theorem 3.6.* Let M be the group of global sections of the vector bundle X over Y = Spec A. This is a finitely generated projective A-module, and  $X = V(M) = \operatorname{Spec} \mathcal{S}_A(M)$ , where  $\mathcal{S}_A(M)$  denotes the symmetric algebra of M over A. Let  $B = A^G$ . Our assumptions imply that  $Y \to Y/\!\!/ G$  is a G-torsor, see Theorem 2.14. Then, A is a faithfully flat extension of B. By assumption, there is a G-equivariant action on the A-module M. This assumption may be translated into descent data on the projective A-module M. Since such descent data are effective, there exists a projective B-module N such that  $A \otimes_B N \cong M$ , and this B-module N is none other but  $M^G$ . One then has  $\mathcal{S}_A(M)^G = \mathcal{S}_B(N)$ . Since  $X/\!\!/ G = \operatorname{Spec} \mathcal{S}_A(M)^G$  and  $Y/\!\!/ G = \operatorname{Spec} B$ , it follows that  $X/\!\!/ G$  is a vector bundle over  $Y/\!\!/ G$ , which is the first assertion. Now note that since all geometric stabilizers of G at a closed point of Y are trivial, certainly the same holds for X, and X is a principal homogeneous space over  $X/\!\!/G$ , just as Y is over  $Y/\!\!/G$ . If Y and hence X are integral, so are  $X/\!\!/ G$  and  $Y/\!\!/ G$ , and  $k(X)^G = k(X/\!\!/ G)$  is purely transcendental over  $k(Y/\!\!/G) = k(Y)^G.$ 

**Corollary 3.8.** Let G be a reductive group over k, let V be a finite-dimensional k-vector space with a linear G-action and let X be an affine k-variety with an almost free G-action. Assume that  $\overline{X}$  is factorial. Then the field  $k(X \times V)^G$  is pure over the field  $k(X)^G$ . In particular, if  $k(X)^G$  is pure, so is  $k(X \times V)^G$ .

Proof. By Corollary 2.19 there exists a non-empty G-invariant affine open set  $U \subset X$  such that  $U \to U/\!\!/ G$  is a principal homogeneous space under G, and  $k(U/\!\!/ G) = k(X)^G$ . The natural projection from  $Z_1 = U \times V$  to Umakes  $Z_1$  into a (trivial) vector bundle over the affine variety U. This vector bundle is equipped with the diagonal G-action. According to Theorem 3.6,  $Z_1/\!\!/ G$  is a vector bundle over  $U/\!\!/ G$ . Since such a bundle is locally trivial in the Zariski topology, the function field  $k(Z_1/\!\!/ G)$  is purely transcendental over  $k(U/\!\!/ G) = k(X)^G$ . On the other hand, for the diagonal action of G on the affine variety  $Z_1 = U \times Y$  all stabilizers are trivial, hence by Luna's Theorem 2.14, the map  $Z_1 \to Z_1/\!\!/ G$  makes  $Z_1$  into a principal homogeneous space over  $Z_1$  under G. In particular,  $k(Z_1/\!\!/ G) = k(Z_1)^G = k(X \times V)^G$ . Thus  $k(X \times V)^G$ is purely transcendental over  $k(X)^G$ .

**Corollary 3.9** (No-name lemma). Let G be a reductive group over k and let V and W be two finite-dimensional k-vector spaces with almost free linear G-actions. Then the fields  $k(V)^G$  and  $k(W)^G$  are stably equivalent. If one of them is pure, the other one is stably pure.

*Proof.* The diagonal action of G on  $V \times W$  gives a common extension:



where purity follows from Corollary 3.8.

**Corollary 3.10.** Let G be a reductive k-group and let V,  $V_1$  and  $V_2$  be finitedimensional k-vector spaces with linear G-actions. Assume that the action on V is almost free. If the action of G on  $V_1$  is almost free and dim  $V_2 \ge$ dim V, then  $k(V_1 \oplus V_2)^G$  is pure over  $k(V)^G$ . If moreover  $k(V)^G$  is pure, so is  $k(V_1 \oplus V_2)^G$ .

*Proof.* Because the action of G on  $V_1$  is almost free, the same argument as in the previous proof shows that  $k(V_1 \oplus V)^G$  is pure over  $k(V_1)^G$ , of transcendence degree dim V. Similarly,  $k(V_1 \oplus V_2)^G$  is pure over  $k(V_1)^G$ , of transcendence degree dim  $V_2$ . Thus  $k(V_1 \oplus V_2)^G$  is pure over  $k(V_1 \oplus V)^G$ . Now since the action of G on V is almost free,  $k(V_1 \oplus V)^G$  is pure over  $k(V)^G$ .  $\Box$ 

**Corollary 3.11.** Let the reductive k-group G act almost freely and linearly on the finite-dimensional k-vector space V. Let  $G \subset \operatorname{GL}_n$  be a closed embedding of G as a subgroup of  $\operatorname{GL}_n$  for some integer n. Then the field  $k(V)^G$  is stably equivalent to the function field  $k(\operatorname{GL}_n/G)$  of the homogeneous space  $\operatorname{GL}_n/G$ .

Proof. Let  $\mathcal{M}_n$  denote be the set of  $n \times n$  matrices. Via the natural (left) multiplication, there is a linear G-action on the vector space  $\mathcal{M}_n$ . On the G-stable affine open set  $U = \operatorname{GL}_n \subset \mathcal{M}_n$ , the group G acts with trivial stabilizers. Thus the action of G on  $\mathcal{M}_n$  is almost free, and Corollary 3.9 implies that  $k(V)^G$  is stably equivalent to the function field  $k(\operatorname{GL}_n)^G$  which, by Remark 2.12, coincides with  $k(\operatorname{GL}_n/G)$ .

See [107, Remark (1.5.7)] for a generalization of the above statement. There also exist projective versions of the no-name method.

**Corollary 3.12.** Let G be a reductive k-group, let V be a finite-dimensional k-vector space with a linear G-action and let X be a geometrically integral k-variety with an almost free G-action. Then the field  $k(X \times \mathbb{P}(V))^G$  is pure over the field  $k(X)^G$ . In particular, if  $k(X)^G$  is pure, so is  $k(X \times \mathbb{P}(V))^G$ .

*Proof.* By restricting X to a suitable open set, we may assume that X is affine and that the action of G on X is free. According to the proof of Theorem 3.6, the projection  $(X \times V)/\!\!/G \to X/\!\!/G$  makes the first space into a vector bundle over the second one (the action on  $X \times V$  being the diagonal one). Since the action of G commutes with that of  $\mathbb{G}_m$  on the factor V, the action of  $\mathbb{G}_m$  descends to an action on  $(X \times V)/\!/G$ , which respects the vector bundle structure. Let  $U \subset X/\!/G$  be an open set over which the vector bundle has a

section, i.e.  $(X \times V) /\!\!/ G \cong U \times V$ . We now have:

$$k(X \times \mathbb{P}(V))^G = k(X \times V)^{\mathbb{G}_m} = k((X \times V)//G)^{G \times \mathbb{G}_m}$$
$$= k(U \times V)^{\mathbb{G}_m} = k(U \times \mathbb{P}(V)) = k((X//G) \times \mathbb{P}(V))$$

and this last field is clearly purely transcendental over  $k(X/\!\!/G)$ .

There also exists a generalized version of the no-name lemma, due to Bogomolov [16], and which he has used to study fields of invariants  $k(V)^G$  when G is a simply connected semisimple group.

3.3. Combining both methods. We now restrict attention to linear actions and prove a basic proposition which uses both the "no-name lemma" and the "slice method".

**Proposition 3.13.** Let G be a reductive k-group and let W be a finitedimensional k-vector space with a linear G-action. Let  $v \in W$  be a point such that:

- (i) the stabilizer of v is a reductive k-group H;
- (ii) the orbit of v under G is dense in W.

Then if V is a finite-dimensional k-vector space with an almost free linear G-action, the action of H on V is almost free and the field  $k(V)^H$  is pure over  $k(V)^G$ . In particular,  $k(V)^H$  is stably pure if and only if  $k(V)^G$  is stably pure.

*Proof.* Consider the diagonal action of G on  $W \times V$ . This is an almost free action. Indeed, let  $U \subset V$  be a G-invariant open set and let  $x \in U$  be such that  $G_x = 1$  and  $G \cdot x$  is closed in U. Then the same properties hold for the open set  $W \times U$  and the point  $y = (0, x) \in V \times U$ . Since the action of G on V is almost free, the no-name lemma implies that  $k(W \oplus V)^G$  is pure over  $k(V)^G$ . On the other hand, the closed subvariety  $Y = \{v\} \times V \subset X = W \times V$ satisfies the assumptions of Theorem 3.1 — the "slice lemma" — for the pair (H,G). Indeed, H is clearly the normalizer of Y and if x and y belong to Y(k)and  $q \in G(\bar{k})$  satisfies qx = y, then  $q \in H(\bar{k})$ . That the orbit of Y under G is dense in  $W \times V$  follows from the density of  $G \cdot x$  in W. Thus by the "slice lemma"  $k(Y)^H$  is isomorphic to  $k(V \oplus W)^G$ . Also, the action of H on Y is isomorphic to the action of H on the vector space V, which is an almost free action since the action of G itself was almost free (use the same open set Uand the same point x as above: since H is closed in G,  $H \cdot x$  is closed in  $G \cdot x$ hence in U). Thus  $k(V)^H$  is pure over  $k(V)^G$ . 

There is a projective variant of the above theorem:

**Proposition 3.14.** Let G be a reductive group over k and let W be a finitedimensional k-vector space with a linear G-action. Let  $v \in W$  be a point such that:

- (i) the stabilizer of v is a reductive group H;
- (ii) the orbit of v under the action of  $G \times \mathbb{G}_m$  is dense in W.

Then if V is a finite-dimensional k-vector space with an almost free linear G-action, the action of H on V is almost free and the field  $k(\mathbb{P}(V))^H$  is stably pure over  $k(\mathbb{P}(V))^G$ .

# 3.4. A criterion for retract rationality.

**Proposition 3.15.** Let G be a reductive group over k. Assume that for any local k-algebra A, with residue field  $\kappa$ , the reduction map on étale cohomology sets  $H^1(A, G) \to H^1(\kappa, G)$  is onto. Let V be a k-vector space with an almost free linear G-action. Then  $k(V)^G$  is the function field of a retract rational k-variety.

*Proof.* One easily checks that if X and Y are two stably k-birationally equivalent geometrically integral k-varieties, then X is retract rational over k if and only if Y is. By the no-name lemma one is therefore reduced to proving that if  $G \subset \operatorname{GL}_n$  is a closed embedding of algebraic groups, and  $X = \operatorname{GL}_n/G$  is the quotient variety, then the variety X is retract rational. For A and  $\kappa$  as above, we have the natural compatible exact sequences of pointed sets

Grothendieck's version of Hilbert's theorem 90 ensures  $H^1(A, \operatorname{GL}_n) = 0$ , hence the map  $X(A) \to H^1(A, G)$  is onto. The first arrow in each sequence is equivariant with respect to the left action of  $\operatorname{GL}_n(A)$ , resp.  $\operatorname{GL}_n(\kappa)$ . Two elements of  $X(\kappa)$  have the same image in  $H^1(\kappa, G)$  if and only if they are in the same orbit of  $G(\kappa)$ . If the map  $H^1(A, G) \to H^1(\kappa, G)$  is onto, one concludes that the reduction map  $X(A) \to X(\kappa)$  is onto. It remains to apply Proposition 1.2.

#### 4. Examples

In this section, k denotes a field of characteristic zero and k an algebraic closure of k.

4.1. Solvable Groups. Details upon the following results may be found in the notes by Kervaire and Vust [72].

**Theorem 4.1** (T. Miyata [93], È. Vinberg [152]). Let the abstract group G act linearly upon the finite dimensional vector space V over k, and assume that there is a complete flag of V which is invariant under the action of G. Then the field  $k(V)^G$  is pure.

See also [150].

*Proof.* The proof is easily reduced to the following nice lemma, whose proof only uses the euclidean division algorithm.  $\Box$ 

**Lemma 4.2.** Let K be a field, and let the abstract group G act upon the polynomial ring in one variable K[t]. If  $K \subset K[t]$  is globally fixed, then there exists a G-invariant polynomial p such that  $K(t)^G = K^G(p)$ .

*Proof.* See [72, II, Lemme 1.1, page 167].

**Proposition 4.3.** Assume  $k = \overline{k}$ . If V is a finite dimensional vector space over k and  $G \subset GL(V)$  is an (abstract) abelian group consisting of semisimple elements, then  $k(V)^G$  is pure.

For G a finite group this is just Fischer's well known theorem [42].

*Proof.* Indeed, all elements of G may be simultaneously diagonalized, so that G actually acts through a maximal torus of GL(V).

**Proposition 4.4.** Assume  $k = \overline{k}$ . If G is a connected solvable group over k and if G acts linearly on the finite dimensional vector space V over k, then  $k(V)^G$  is pure.

*Proof.* This is a consequence of the above theorem and of the Lie-Kolchin theorem.  $\hfill \Box$ 

If the (abstract group) G acts linearly on the vector space V, this action induces an action on the projective space  $\mathbb{P}(V)$ .

**Proposition 4.5.** The field  $k(V)^G$  is pure over  $k(\mathbb{P}(V))^G$ .

*Proof.* This follows from Lemma 4.2. See [72, II, Corollaire 1.4, page 168].  $\Box$ 

For more general actions, we have the following

**Theorem 4.6** (Rosenlicht [114, 116]). Let G be a connected solvable algebraic group over an algebraically closed field k. If X is an integral variety with a G-action, and  $f: X \to Y$  is a geometric quotient for this action, then f admits a section over a non-empty open set of Y.

This theorem has the following consequences:

- (i) If X is a rational variety, then Y is a retract rational variety.
- (ii) If the action of G on X is free, i.e. if X is a G-torsor over Y, then X is birational to  $G \times Y$ ; if moreover X is rational, then Y is stably rational.

4.2. **Special Groups.** Recall that an algebraic group G over a field k is called *special* if any principal homogeneous space (torsor) under G over a k-variety is locally trivial for the Zariski topology. This implies  $H^1(K, G) = 0$  that for any field K containing k.

The following facts were proved by Serre [28]. Any special group G is linear and connected. If  $G \subset \operatorname{GL}_n$  is some closed embedding, G is special if and only if the fibration  $\operatorname{GL}_n \to \operatorname{GL}_n / G$  is locally trivial for the Zariski topology.

Special groups include

- (i) the additive group  $\mathbb{G}_a$ ,
- (ii) the multiplicative group  $\mathbb{G}_m$ ,
- (iii) more generally, split connected solvable groups,

as well as some connected linear algebraic groups, e.g.,

(v) the general linear groups  $\operatorname{GL}_n$ , more generally,  $\operatorname{GL}(A)$  for A a central simple algebra over k,

- (vi) the special linear groups  $SL_n$ ,
- (vii) the symplectic groups  $Sp_{2n}$ ,
- (viii) the split spinor groups  $\operatorname{Spin}_n$  for  $n \leq 6$ .

This last case follows from the existence of the exceptional isomorphisms in the classification of Lie groups:  $\text{Spin}_3 \cong \text{SL}_2$ ,  $\text{Spin}_4 \cong \text{SL}_2 \times \text{SL}_2$ ,  $\text{Spin}_5 \cong \text{Sp}_4$ ,  $\text{Spin}_6 \cong \text{SL}_4$ . The special orthogonal groups  $\text{SO}_n$  for  $n \geq 3$  and the spinor groups for  $n \geq 7$  are not special.

Over an algebraically closed field, any connected linear algebraic group is a rational variety. Over an arbitrary field k, if a semisimple group is special, it is a product of copies of Weil restriction of scalars of groups of type  $SL_n$ and  $Sp_{2n}$  (Grothendieck [28] over an algebraically closed field, one checks this extends to arbitrary k). Thus the underlying variety of a special semisimple k-group is a k-rational variety. This statement is not true for special k-tori.

**Proposition 4.7.** Let G be a special group over k whose underlying variety is k-rational, and let X be a factorial affine k-variety with an almost free Gaction. Then k(X) is purely transcendental over  $k(X)^G$ . In particular, if X is a (stably) rational variety, then  $k(X)^G$  is stably pure.

Proof. Since the action is almost free, there exist a non-empty open set  $U \subset X$ and a morphism  $U \to Y$  which makes U into a G-torsor over Y (Corollary 2.19). In particular, the function field k(Y) coincides with  $k(X)^G$ . Because G is special, any G-torsor is locally trivial for the Zariski topology and U, hence also X, is birational to the product  $Y \times G$ . Thus k(X) is pure over  $k(Y) = k(X)^G$ .

**Corollary 4.8.** Let G be a semisimple special group over k, and let V be a finite dimensional k-vector space equipped with an almost free linear G-action. Then  $k(V)^G$  is stably pure.

*Proof.* Indeed, as mentioned above, G is a k-rational variety.

**Proposition 4.9.** Let G be a linear algebraic group over k. Let  $G_1, G_2$  be two special groups which contain G as a closed subgroup. Assume that the underlying varieties of  $G_1$  and  $G_2$  are rational over k. Then  $G_1/G$  and  $G_2/G$ are stably k-birationally equivalent.

Proof. Let  $X = (G_1 \times_k G_2)/G$  be the quotient of  $G_1 \times_k G_2$  under the diagonal embedding. The projection map  $X \to G_1/G$  makes X into a  $G_2$ -torsor over  $G_1/G$ . Since  $G_2$  is special, X is k-birational to the product  $G_2 \times (G_1/G)$ . Thus X is stably k-birational to  $G_1/G$ . The same argument applies to the projection  $X \to G_2/G$ . Hence  $G_1/G$  is stably k-birationally equivalent to  $G_2/G$ .

Example 4.10. Let  $G \subset \operatorname{GL}_p$  and  $G \subset \operatorname{SL}_q$  be two algebraic k-group embeddings of a given algebraic group G. Then the function fields  $k(\operatorname{GL}_p/G)$  and  $k(\operatorname{SL}_q/G)$  are stably equivalent.

## 4.3. The Orthogonal Group.

**Proposition 4.11.** Let q be a nondegenerate quadratic form over k of rank  $n \geq 2$ . Let G = O(q) be the orthogonal group of q. Let V be a finitedimensional k-vector space with an almost free linear G-action. Then the field  $k(V)^G$  is stably pure over k.

1st Proof (by reduction to  $\operatorname{GL}_n$ ). Apply Proposition 3.13 to the group  $G = \operatorname{GL}_n$  acting upon the vector space V of symmetric matrices M through  $g \cdot M = {}^t g M g$ , and take v in that proposition to be the matrix S of q. The orbit of v is the open set of non-singular symmetric matrices, hence it is dense. The stabilizer of v is O(q). Then Proposition 3.13 reduces the case of O(q) to that of  $\operatorname{GL}_n$  which is a "special" group. Then we are done by Proposition 4.7.  $\Box$ 

2nd Proof. Let  $Q_n \subset \mathcal{M}_n$  be the vector space consisting of symmetric matrices. Consider the map  $\mathcal{M}_n \to Q_n$  which sends the matrix A to  ${}^tASA$ . This map induces a map f of affine varieties  $X = \operatorname{GL}_n \to Y = Q_n^{\operatorname{ns}}$ , where  $Q_n^{\operatorname{ns}}$  denotes the open set of non-singular symmetric  $n \times n$  matrices. Let G = O(q) act upon  $\mathcal{M}_n$ , hence also  $\operatorname{GL}_n$ , by left translation, i.e. by the action  $(g, A) \mapsto gA$ . One easily checks that f makes X into a G-torsor over Y for this action, hence  $k(X)^G = k(Y)$  (that the map f is onto over  $\bar{k}$  reflects the fact that any nondegenerate quadratic form of rank n is equivalent to q over  $\bar{k}$ ). Since  $Y = Q_n^{\operatorname{ns}}$ is an open set in a vector space, k(Y) is purely transcendental. On the other hand, the action of G on  $\mathcal{M}_n$  is linear, and it is an almost free action. The theorem now follows from the no-name lemma (Corollary 3.9).

Remark 4.12. Using Proposition 1.2, the retract rationality of  $k(V)^G$  can already be seen from the following fact: For any local k-algebra A with residue class field  $\kappa$ , the natural map  $H^1(A, O(q)) \to H^1(\kappa, O(q))$  is surjective. Indeed,  $H^1(\kappa, O(q))$  classifies isomorphism classes of non-degenerate rank n quadratic forms over  $\kappa$ . Any such class  $\alpha$  contains a diagonal form  $a_1x_1^2 + \cdots + a_nx_n^2$ . Lifting the  $a_i \in \kappa^*$  to elements of  $A^*$  produces a non-degenerate quadratic form, hence an element in  $H^1(A, O(q))$  whose image under the restriction map is  $\alpha$ .

4.4. The Special Orthogonal Group. Let q be a nondegenerate quadratic form over k of rank  $n \geq 2$ , let  $\operatorname{disc}(q) \in k^*/k^{*2}$  be its discriminant. Let  $G = \operatorname{SO}(q)$  be the special orthogonal group of q. If A is a local k-algebra with  $2 \in A^*$ , then  $H^1(A, \operatorname{SO}(q))$  classifies non-degenerate quadratic forms over A with the same discriminant as q in  $A^*/A^{*2}$ . Now, if A is a local ring with residue class field  $\kappa$  (char  $\kappa \neq 2$ ) and  $\sum a_i x_i^2$  is a non-degenerate quadratic form over  $\kappa$  with discriminant disc(q), it can clearly be lifted to a non-degenerate quadratic form over A with the same property.

Just as in Remark 4.12, the lifting property just proven implies that for an almost free linear G-action of G = SO(q) on a finite dimensional vector space V over k the field  $k(V)^G$  is retract rational. One can actually do better.

**Proposition 4.13.** Let q be a nondegenerate quadratic form over k of rank  $n \geq 2$ . For any almost free finite dimensional linear representation V of the special orthogonal group SO(q) over k, the field  $k(V)^{SO(q)}$  is stably rational.

Proof. Let  $X = V^{n-1}$ , let  $Y = k^N$ , N = n(n-1)/2, and let  $f: X \to Y$ be the map which sends  $(e_1, \ldots, e_{n-1})$  to the point  $\{\langle e_i, e_j \rangle\}_{1 \le i \le j \le n-1}$ , where  $\langle x, y \rangle$  denotes the "scalar product" of two vectors x and y with respect to the bilinear form assolated to q. Applying Proposition 2.4 to the open set U of X consisting of  $(e_1, \ldots, e_{n-1})$  with  $e_1 \wedge \cdots \wedge e_{n-1} \neq 0$ , one obtains  $k(Y) = k(X)^G$ . Moreover, the action of G on X is almost free. Since Y is rational, this produces an almost-free finite dimensional linear representation X over k such that  $k(X)^G$  is pure over k. We conclude by the no-name lemma (Corollary 3.9).

*Remark* 4.14. To prove this result, Bogomolov [16] uses a generalized version of the no-name lemma together with the slice method; see also Saltman's approach [124].

4.5. The Spinor Group. Let Spin(q) denote the spinor group attached to a nondegenerate quadratic form q of rank n. If  $k = \bar{k}$ , we write simply  $\text{Spin}(q) = \text{Spin}_n$ .

Assume  $k = \bar{k}$ . As we saw in §4.2 for  $n \leq 6$ , the group  $\operatorname{Spin}_n$  is special, hence if V is an almost free finite dimensional linear representation of  $\operatorname{Spin}_n$ , the field  $k(V)^{\operatorname{Spin}_n}$  is stably pure. In the literature, there is a proof [16] of the stable purity of  $k(V)^{\operatorname{Spin}_n}$  for arbitrary n, but that proof is incomplete, it builds upon an incorrectly proved lemma [16, Lemma 3.1]. A proof of the stable rationality of  $\operatorname{Spin}_n$  for n = 7 and n = 10 is given by V.È. Kordonskiĭ [80].

D. Shapiro suggested that the classification of forms of low degree given by A. Pfister in 1966 would immediately yield the retract rationality of  $k(V)^{\text{Spin}_n}$  for  $n \leq 12$ . Here we shall only discuss the case n = 12.

**Theorem 4.15.** Let  $G = \text{Spin}(q_0)$  be the spinor group of the nondegenerate hyperbolic quadratic form  $q_0$  of rank 12. Let V be a finite-dimensional k-vector space with an almost free linear G-action. Then the field  $k(V)^G$  is retract rational over k.

*Proof.* Let us write  $\text{Spin} = \text{Spin}(q_0)$  and  $\text{SO} = \text{SO}(q_0)$ . Let

$$1 \rightarrow \mu_2 \rightarrow \text{Spin} \rightarrow \text{SO} \rightarrow 1$$

be the natural sequence. For any k-algebra A, there is an an exact sequence of pointed sets in étale cohomology:

$$H^1(A, \mu_2) \to H^1(A, \operatorname{Spin}) \to H^1(A, \operatorname{SO}) \xrightarrow{\mathcal{O}} H^2(A, \mu_2)$$

There is an action of the group  $H^1(A, \mu_2)$  on  $H^1(A, \operatorname{Spin})$ , and this action is transitive on elements of  $H^1(A, \operatorname{Spin})$  with the same image in  $H^1(A, \operatorname{SO})$  [28; 137, I. 5.7, Prop. 42 p. 52]. The set  $H^1(A, \operatorname{SO})$  classifies the non-degenerate quadratic forms over A of dimension n = 12 with discriminant  $1 \in A^*/A^{*2}$ . The map

$$H^1(A, \mathrm{SO}) \xrightarrow{\partial} H^2(A, \mu_2)$$

associates to the class of a such a form its Clifford invariant [74, p. 437; 137, pp. 147–148].

Let A be a local k-algebra, let  $\kappa$  be its residue field. Let  $\xi \in H^1(\kappa, \operatorname{Spin})$ . Its image in  $H^1(\kappa, \operatorname{SO})$  is the isomorphy class of a quadratic form q of rank 12, trivial discriminant, and trivial Clifford invariant. According to a theorem of Pfister [101, Satz 14] (see also [59]) there exist elements a, b, c, d, e, f in  $k^*$  such that such a form may be written as  $q = a\langle 1, b \rangle \otimes \langle -c, -d, cd, e, f, -ef \rangle$ . Since the reduction map  $A^*/A^{*2} \to \kappa^*/\kappa^{*2}$  is onto, one may lift this quadratic form to a class of the same shape over A, and the Clifford invariant of such a class is trivial, as may be checked by a direct computation. Indeed, such a class clearly lies in  $I^3A$ . One thus finds an element  $\eta_A \in H^1(\Lambda, \operatorname{Spin})$  whose image  $\eta_{\kappa} \in H^1(\kappa, \operatorname{Spin})$  has the same image as  $\xi$  in  $H^1(\kappa, \operatorname{Spin})$ . There thus exists  $\rho_{\kappa} \in H^1(\kappa, \mu_2)$  such that  $\rho_{\kappa}.\eta_{\kappa} = \xi$ . Let  $\rho \in H^1(\Lambda, \mu_2)$  be a lift of this element. Then  $\rho \cdot \eta \in H^1(\Lambda, \operatorname{Spin})$  reduces to  $\xi \in H^1(\kappa, \operatorname{Spin})$ .

4.6. The Projective Linear Group. Let V be a finite dimensional vector space over  $k, n \geq 2$  an integer. Suppose the projective linear group PGL<sub>n</sub> acts linearly and almost freely on V. The question whether the field of invariants  $k(V)^{\text{PGL}_n}$  is rational (pure), or at least stably rational, has come up in a variety of contexts. It is still open.

The simplest linear representation of the above type is given by  $V_n = \mathcal{M}_n(k) \oplus \mathcal{M}_n(k)$  with PGL<sub>n</sub> acting by simultaneous conjugation. The field of invariants  $C_n := k(V_n)^{\text{PGL}_n}$  coincides with the field of fractions

The field of invariants  $C_n := k(V_n)^{\text{PGL}_n}$  coincides with the field of fractions of the centre of the generic division ring on two  $n \times n$  generic matrices (Procesi [109, pp. 240–241]; M. Artin [2]).

At least for k the field of complex numbers, the field  $C_n$  may also be viewed as

- The function field of the moduli space M(n; 0, n) of stable rank n vector bundles on  $\mathbb{P}_2$  with Chern numbers  $c_1 = 0$ ,  $c_2 = n$  (see Le Bruyn [84,85]).
- The function field of the generic Jacobian variety of plane curves of degree n (Van den Bergh [151]).

Katsylo and Schofield proved the following reduction result (see also [127]):

**Proposition 4.16** ([66,135]). Suppose n = rs with gcd(r, s) = 1, then  $C_n$  is stably equivalent to the fraction field of  $C_r \otimes C_s$ .

Saltman proved retract rationality of  $C_n$  for n prime [118, Corollary 5.3]. For an alternate proof, see also [37, Corollary 9.13].

**Proposition 4.17.** Let n be squarefree or twice a squarefree number. Then for any almost free finite dimensional linear representation V of the projective linear group group  $\operatorname{PGL}_n$ , the field  $k(V)^{\operatorname{PGL}_n}$  is retract rational.

The rationality for n = 2 was proved by Sylvester [148], and by Procesi [109]. Formanek solved the rationality problem for n = 3 [43] and n = 4 [44]. For n arbitrary, Formanek [43] gave a very useful description of  $C_n$  as a field of multiplicative invariants under an action of the symmetric group. For  $k = \bar{k}$ , Bessenrodt and Le Bruyn [12] proved the stable rationality for n = 5, 7. A simpler proof was later given by Beneish [7]. The question remains open for all other prime powers n. The quoted results combine to :

**Proposition 4.18.** Let  $k = \bar{k}$ . Let n = 2, 3, 4, 5, 7, or more generally let n be any divisor of 420. Then for any almost free linear representation V of the projective linear group group  $PGL_n$ , the field  $k(V)^{PGL_n}$  is stably rational.

For further rationality results on  $k(V)^{\text{PGL}_n}$ , with connections with the rationality of some moduli varieties, see [15, 20, 22, 61-71, 140-143]. The case of some exceptional groups is considered in [54, 55].

### 4.7. Some Finite Groups.

**Lemma 4.19.** Let X be a geometrically integral k-variety, and G a finite group acting on X. If the action of G is faithful, then the action is almost free.

*Proof.* For any  $g \in G$ ,  $g \neq e$ , the k-variety  $X^g \subset X$  of fixed points has codimension at least 1. Let U be the complement of the union of the  $X^g$ ,  $g \neq e$ . Then U is a G-stable nonempty open set and the action of G on U is free.

**Proposition 4.20.** For any faithful finite dimensional linear representation V of  $\mathfrak{S}_n$  over k, the field  $k(V)^{\mathfrak{S}_n}/k$  is stably pure over k.

*Proof.* This is the most classical case. For the natural linear representation of  $\mathfrak{S}_n$  in  $k^n$  by permutations on the canonical basis, the theorem on symmetric polynomials says that

$$k(t_1,\ldots,t_n)^{\mathfrak{s}_n} = k(s_1,\ldots,s_n)$$

is a pure extension of k. Here we denote by  $s_1 = t_1 + \cdots + t_n, \ldots, s_n = t_1 \ldots t_n$  the fundamental symmetric polynomials. The proposition now follows from the no-name lemma.

But the analogous question for  $\mathfrak{A}_n$ ,  $n \geq 6$  remains open!

**Proposition 4.21.** Assume  $k = \bar{k}$ . Let G be a finite subgroup of GL(3, k). Then for any faithful finite dimensional linear representation V of G over k, the field  $k(V)^G/k$  is stably pure over k.

*Proof.* Let  $V_0$  be a faithful representation, dim  $V \leq 3$ . By Proposition 4.5 we know that the field  $k(V_0)^G$  is pure over the field  $k(\mathbb{P}(V_0))^G$ . Since this last field is unirational of transcendence degree at most 2, it is purely transcendental by Castelnuovo's theorem. The result now follows from the no-name lemma.  $\Box$ 

Remark 4.22. Let k be a field and  $G \subset \operatorname{GL}(2, k)$  be any finite subgroup. Then for any faithful finite dimensional linear representation V of G over k, the field  $k(V)^G/k$  is stably pure. This result applies to the dihedral group  $G = D_n$  of order 2n and  $k = \bar{k}$  or  $k = \mathbb{R}$ . For the proof we consider a faithful representation  $V_0$ , dim  $V_0 \leq 2$ . By Proposition 4.5 and Lüroth's theorem,  $k(V_0)^G/k$  is pure. Hence  $k(V)^G/k$  is stably pure by the no-name lemma.

In some cases, one may use the no-name lemma to prove the purity of fields of invariants.

**Theorem 4.23** (Maeda [90]). Let V be the natural 5-dimensional representation of the alternating group  $\mathfrak{A}_5$ . Then  $k(V)^{\mathfrak{A}_5}$  is pure over k.

We here give a proof in the case  $k = \bar{k}$ .

Proof (for k = k). There exists a faithful representation W of dimension 3 (as the group of automorphisms of the icosahedron). Also, there is a decomposition  $V = V_0 \oplus V_1$  where  $V_0$  is the trivial representation. Now  $V/\mathfrak{A}_5$  is pure over  $\mathbb{P}(V_1)/\mathfrak{A}_5$  and of transcendence degree 2. On the other hand, the action of  $\mathfrak{A}_5$  on  $\mathbb{P}(V_1)$  and  $\mathbb{P}(W)$  is almost free (indeed,  $\mathfrak{A}_5$  is simple). We may thus apply the projective no-name lemma, i.e. Corollary 3.12. This implies that  $(\mathbb{P}(V_1) \times \mathbb{P}(W))/\mathfrak{A}_5$  is pure over  $\mathbb{P}(V_1)/\mathfrak{A}_5$  of transcendence degree 2. Thus  $\mathbb{P}(V_1)/\mathfrak{A}_5$  and  $(\mathbb{P}(V_1) \times \mathbb{P}(W))/\mathfrak{A}_5$  are birational. A second application of the projective no-name lemma implies that  $(\mathbb{P}(V_1) \times \mathbb{P}(W))/\mathfrak{A}_5$  is pure over  $\mathbb{P}(W)/\mathfrak{A}_5$ , which is unirational of dimension 2, hence pure by Castelnuovo's theorem since char k = 0.

We refer to [90] for the proof over  $\mathbb{Q}$ . In [102], B. Plans shows that for odd  $n \geq 3$ , the field  $\mathbb{Q}(X_1, \ldots, X_n)^{\mathfrak{A}_n}$  is pure over  $\mathbb{Q}(X_1, \ldots, X_{n-1})^{\mathfrak{A}_{n-1}}$ , thus giving another proof of Maeda's theorem.

Remark 4.24. One may also prove that  $k(\mathbb{P}(V))^{\mathfrak{A}_5}$  is pure, see [52, Lemma 5].

Remark 4.25. The binary icosahedral group is a subgroup of  $SL_2(\mathbb{C})$ . It is a double cover of  $\mathfrak{A}_5$ , actually it is its "representation group" in the sense of Schur. It has a unique 4-dimensional faithful linear action. The analysis of the quotient requires a much more delicate analysis [77]. See also [78, 79] for some other finite subgroups of GL<sub>4</sub>.

For further results on the rationality of the field of invariants for a linear action of other small finite groups, see [1, 29, 30, 111, 123].

### 5. The unramified Brauer group

In this section, we shall assume some knowledge of group cohomology, as well as some knowledge of local fields, all of which may be found in Serre's book [136]. For the sake of simplicity, all fields  $(k, K, L, \kappa, ...)$  will be taken of characteristic zero. Let K be a field, let  $\overline{K}$  be an algebraic closure of K and let  $\mathfrak{g} = \operatorname{Gal}(\overline{K}/K)$ .

**Definition 5.1.** The second (profinite) cohomology group  $H^2(\mathfrak{g}, \overline{K}^*)$  of  $\mathfrak{g}$  with values in the multiplicative group  $\overline{K}^*$  of  $\overline{K}$  is called the Brauer group of K and is denoted Br K.

Given any field inclusion  $K \subset L$ , there is a natural map  $\operatorname{Br} K \to \operatorname{Br} L$ .

**Definition 5.2.** When K is the field of fractions of a discrete valuation ring A with residue field  $\kappa$  (of characteristic zero), there is a basic homomorphism

$$\partial_A \colon \operatorname{Br} K \to X(\kappa),$$

where  $X(\kappa)$  denotes the group  $\operatorname{Hom}_{\operatorname{cont}}(\operatorname{Gal}(\bar{\kappa}/\kappa), \mathbb{Q}/\mathbb{Z})$  of continuous characters of  $\operatorname{Gal}(\bar{\kappa}/\kappa)$  with values in the discrete group  $\mathbb{Q}/\mathbb{Z}$ .

Let us first assume that A is complete. Since the characteristic of  $\kappa$  is assumed to be zero, A, resp. K, may be then identified with the ring of power series  $\kappa[[t]]$ , resp. with its fraction field  $\kappa((t))$ . Let  $K_{nr}$  be the maximal unramified extension of K. Under the previous identifications,  $K_{nr}$  coincides with  $\overline{k}((t))_{alg}$  (the algebraic closure of  $\kappa((t))$  in  $\overline{k}((t))$ ),  $\operatorname{Gal}(K_{nr}/K) = \operatorname{Gal}(\overline{\kappa}/\kappa)$ , and  $\overline{K}$  is the union of all fields  $\overline{\kappa}((t^{1/n}))_{alg}$  ("Puiseux's theorem"). The Galois group  $\operatorname{Gal}(\overline{K}/K_{nr})$  can thus be identified with the profinite group  $\varprojlim \mu_n$ , which in a non canonical manner is isomorphic to the group  $\widehat{\mathbb{Z}} = \varprojlim \mathbb{Z}/n$ . The cohomological dimension of such a group is 1, hence  $H^2(\operatorname{Gal}(\overline{K}/K_{nr}), \overline{K}^*) = 0$ . Since also  $H^1(\operatorname{Gal}(\overline{K}/K_{nr}), \overline{K}^*) = 0$  according to Hilbert's Theorem 90, the restriction-inflation sequence gives rise to a natural isomorphism:

$$H^{2}(\operatorname{Gal}(K_{\operatorname{nr}}/K), K_{\operatorname{nr}}^{*}) \xrightarrow{\cong} H^{2}(\operatorname{Gal}(\overline{K}/K), \overline{K}^{*}) = \operatorname{Br} K.$$
 (5.1)

The valuation  $v: K^* \to \mathbb{Z}$  naturally extends to a valuation  $v: K_{nr}^* \to \mathbb{Z}$  which respects the action of  $\operatorname{Gal}(K_{nr}/K)$  ( $\mathbb{Z}$  being taken with trivial action). It therefore induces a homomorphism:

$$H^{2}(\operatorname{Gal}(K_{\operatorname{nr}}/K), K_{\operatorname{nr}}^{*}) \to H^{2}(\operatorname{Gal}(K_{\operatorname{nr}}/K), \mathbb{Z}) = H^{2}(\operatorname{Gal}(\bar{\kappa}/\kappa), \mathbb{Z}).$$
(5.2)

Now the cohomology of the exact sequence  $0 \to \mathbb{Z} \to \mathbb{Q} \to \mathbb{Q}/\mathbb{Z} \to 0$  identifies the last group with  $H^1(\operatorname{Gal}(\bar{\kappa}/\kappa), \mathbb{Q}/\mathbb{Z}) = \operatorname{Hom}_{\operatorname{cont}}(\operatorname{Gal}(\bar{\kappa}/\kappa), \mathbb{Q}/\mathbb{Z})$ . Combining the isomorphism (5.1) with the map (5.2) defines the map  $\partial_A$  in the case when A is complete.

In the general case,  $\partial_A$  will be defined as the composite map

$$\operatorname{Br} K \to \operatorname{Br} \widehat{K} \xrightarrow{\mathcal{O}_{\widehat{A}}} X(\kappa)$$

where  $\widehat{A}$ , resp.  $\widehat{K}$ , are the completions of A, resp. K — the residue class field  $\kappa$  being the same as that of A.

We are now in a position to define our basic invariant, first used efficiently by Saltman [119].

**Definition 5.3.** Let k be a field of characteristic zero. Let K/k be a function field, i.e. assume that the field K is finitely generated, as a field, over the field k. The subgroup  $\bigcap_A \ker \partial_A$  of Br K, where A runs through all discrete rank one valuation rings  $A \subset K$  such that the field of fractions of A is K and k is included in A, is called the *unramified Brauer group* of K (with respect to k) and it is denoted  $\operatorname{Br}_{nr}(K/k)$ , or  $\operatorname{Br}_{nr} K$  if there is no ambiguity on k.

*Remark.* In the case where k is algebraically closed,  $k^*$  is infinitely divisible, hence for any discrete valuation ring  $A \subset K$  with fraction field K, we have  $k \subset A$ . Hence, in this case, we just write  $\operatorname{Br}_{\operatorname{nr}} K$ .

**Lemma 5.4.** The natural map  $\operatorname{Br} k \to \operatorname{Br} K$  sends  $\operatorname{Br} k$  to  $\operatorname{Br}_{\operatorname{nr}}(K/k)$ .

*Proof.* It is enough to prove the statement when K is the field of fractions of a complete discrete valuation ring A containing k, i.e.  $A = \kappa[[t]]$  and  $K = \kappa((t))$ , with  $k \subset \kappa$ . The composite map

$$k^* \to \bar{\kappa}((t))^*_{\text{alg}} = K^*_{\text{nr}} \to \mathbb{Z},$$

where the last map is given by the valuation, is zero. Hence the composite map

$$H^{2}(\operatorname{Gal}(\bar{k}/k), \bar{k}^{*}) \to H^{2}(\operatorname{Gal}(K_{\operatorname{nr}}/K), K_{\operatorname{nr}}^{*}) \to H^{2}(\operatorname{Gal}(\bar{\kappa}/\kappa), \mathbb{Z}) = X(\kappa)$$
  
is zero.

**Lemma 5.5.** Let  $K \subset L$  be function fields over the field k. The natural map Br  $K \to$  Br L induces a map Br<sub>nr</sub>(K/k)  $\to$  Br<sub>nr</sub>(L/k).

Proof. Let  $\alpha$  be an element of  $\operatorname{Br}_{\operatorname{nr}}(K/k)$ , and let  $\alpha_L$  be its image in  $\operatorname{Br} L$ . Let  $B \subset L$  be a discrete valuation ring of rank one with field of fractions Land with  $k \subset L$ . Let  $A = B \cap K$  be the trace of A on K. We have  $k \subset A$ . If A = K, then  $\partial_B(\alpha_L) = 0$  according to the previous lemma. Otherwise A is a discrete valuation ring of rank one with field of fractions K, and there is a natural inclusion of the residue class field  $\kappa_A$  of A into the residue class field  $\kappa_B$  of B. Let  $\pi$  be a uniformizing parameter of A, and let  $e = v_B(\pi) > 0$  be the valuation of  $\pi$  in B. The result now follows from the general fact that in such a situation, there is a commutative diagram:

$$\begin{array}{ccc} \operatorname{Br} L & \stackrel{\partial_B}{\longrightarrow} & X(\kappa_B) \\ \operatorname{Res}_{K/L} & & \uparrow^{e \cdot \operatorname{Res}_{\kappa_A/\kappa_B}} \\ \operatorname{Br} K & \stackrel{\partial_A}{\longrightarrow} & X(\kappa_A) \end{array}$$
(5.3)

where the L.H.S. vertical map is the natural map, and the R.H.S. vertical map is *e* times the map induced on character groups by the inclusion  $\kappa_A \subset \kappa_B$ . To prove this last fact, one simply goes over to completions, in which case the inclusion  $K \subset L$  (resp.  $K_{nr} \subset L_{nr}$ ) reads  $\kappa_A((t)) \subset \kappa_B((u))$  (resp.  $\bar{\kappa}_A((t))_{alg} \subset \bar{\kappa}_B((u))_{alg}$ ) with  $u = \rho t^e$  for some  $\rho \in \kappa_B^*$ .

**Lemma 5.6.** If K = k(t) is the rational field in one variable over the field k, the natural map  $Br(k) \rightarrow Br_{nr}(K/k)$  is an isomorphism.

*Proof.* Let k be an algebraic closure of k. Let G = Gal(k/k). Since k[t] is a unique factorization domain and its units are reduced to  $\bar{k}^*$ , there is an obvious exact sequence of G-modules:

$$1 \to \bar{k}^* \to \bar{k}(t)^* \xrightarrow{\{v_P\}} \oplus_P \mathbb{Z}_P \to 0.$$
(5.4)

Here P runs through all monic irreducible polynomials in k[t]. If P is such a polynomial, and  $P(t) = (t - \alpha_1) \dots (t - \alpha_n)$  is its decomposition over  $\bar{k}$ , one lets  $\mathbb{Z}_P = \mathbb{Z}\alpha_1 \oplus \dots \oplus \mathbb{Z}\alpha_n$  be the permutation G-lattice (free  $\mathbb{Z}$ -module) with  $\mathbb{Z}$ -basis the roots  $\{\alpha_1, \dots, \alpha_n\}$  of P in  $\bar{k}$ , the action of G being given by the permutation action on these roots. The map  $\{v_P\}$  sends any function to its divisor, i.e. to the set of zeros and poles with multiplicities.

Sequence (5.4) is a split G-sequence: indeed, the map which associates to  $f \in \bar{k}(t)^*$  the value at zero of  $f/t^{v(f)}$ , where v is the valuation associated to the polynomial t, defines a G-retraction of the embedding  $k^* \to \bar{k}(t)^*$ . Thus the G-cohomology of sequence (5.4) gives rise to a (split) short exact sequence

$$0 \to H^2(G, \bar{k}^*) \to H^2(G, \bar{k}(t)^*) \xrightarrow{\{v_P\}} \oplus_P H^2(G, \mathbb{Z}_P) \to 0.$$
 (5.5)

If 
$$k(\alpha) = k[t]/P$$
, the group  $H^2(G, \mathbb{Z}_P)$  may be identified with  
 $H^2(\operatorname{Gal}(\bar{k}/k(\alpha), \mathbb{Z}) = H^1(\operatorname{Gal}(\bar{k}/k(\alpha), \mathbb{Q}/\mathbb{Z}) = X(k(\alpha)).$ 

Tsen's theorem asserts that the field  $\bar{k}(t)$  is a  $C_1$ -field, and, as such, its Brauer group is trivial. Using Hilbert's theorem 90 and the restriction-inflation sequence we may conclude that the inflation map is an isomorphism

$$H^2(G, \bar{k}(t)^*) \xrightarrow{\cong} Br(k(t)).$$

Let now P be a monic irreducible polynomial in k[t]. Such a polynomial defines a valuation on k(t). Let A be the associated discrete valuation ring, whose residue class field may be identified with the field  $k(\alpha) = k[t]/P$ . By comparing with the above definition of  $\partial_A$ , one then easily checks that the following diagram is commutative:

$$\begin{array}{ccc} H^2(G,\bar{k}(t)^*) & \xrightarrow{v_P} & H^2(G,\mathbb{Z}_P) \\ \cong & & & \downarrow \cong \\ \operatorname{Br}(k(t)) & \xrightarrow{\partial_A} & X(k(\alpha)). \end{array}$$

Putting this together with sequence (5.5) completes the proof.

**Proposition 5.7.** Let K be a function field over the field k. Let t be an indeterminate. The natural map from Br K to Br(K(t)) induces an isomorphism Br<sub>nr</sub>(K/k)  $\cong$  Br<sub>nr</sub>(K(t)/k). In particular, if K/k is rational, i.e. if K is purely transcendental over k, or more generally if K/k is stably rational, i.e. if K(t<sub>1</sub>,...,t<sub>r</sub>) is purely transcendental over k for some suitable independent variables  $t_1, \ldots, t_r$ , then the natural map from Br(k) to Br<sub>nr</sub>(K/k) is an isomorphism.

*Proof.* It is enough to prove the first assertion. Let L = K(t). According to Lemma 5.6 the map  $\operatorname{Br} K \to \operatorname{Br}(K(t))$  is an injection and according to Lemma 5.5 it induces an inclusion  $\operatorname{Br}_{\operatorname{nr}}(K/k) \subset \operatorname{Br}_{\operatorname{nr}}(L/k)$ . By the very definition of the unramified Brauer group, any  $\alpha$  in  $\operatorname{Br}_{nr}(L/k)$  clearly belongs to  $\operatorname{Br}_{\operatorname{nr}}(L/K) = \operatorname{Br}_{\operatorname{nr}}(K(t)/K)$ , and Lemma 5.6 says that this last group coincides with Br K. Thus we only have to prove that if  $\alpha \in Br K$  becomes unramified over k when viewed in Br(K(t)), then it already belongs to  $Br_{nr}(K/k)$ . Let A be a discrete rank one valuation ring with field of fractions K. The localization of A[t] at the prime ideal of A[t] spanned by a uniformizing parameter  $\pi$  of A is a discrete valuation ring B with uniformizing parameter  $\pi$ , and the induced map on residue class fields  $\kappa_A \to \kappa_B$  may be identified with the inclusion  $\kappa_A \to \kappa_A(t)$ . Hence in diagram (5.3) applied to the present situation, the R.H.S. vertical map, i.e.,  $\operatorname{Res}_{\kappa_A/\kappa_B} \colon X(\kappa_A) \to X(\kappa_B) = X(\kappa_A(t))$ is an injection (this applies more generally as soon as e = 1 and  $\kappa_A$  is algebraically closed in  $\kappa_B$ ). Thus for  $a \in \operatorname{Br} K$ , the equality  $\partial_B(\alpha_L) = 0$  implies  $\partial_A(\alpha) = 0$ . Since A was an arbitrary rank one discrete valuation ring in K, the conclusion follows. 

*Remark* 5.8. More generally, one may show that if K is the function field of a k-variety which is retract rational (over k), then the natural map from Br(k)

to  $\operatorname{Br}_{\operatorname{nr}}(K/k)$  is an isomorphism. The proof requires functorial properties of the unramified Brauer group more elaborate than the one given in Lemma 5.5. For this, we refer to [32].

Up till now, we have sticked to a down-to-earth definition of the unramified Brauer group. In some circumstances it is necessary to use a high-brow definition.

**Definition 5.9** (Grothendieck [48]). The Brauer group Br X of a scheme X is the second étale cohomology group  $H^2_{\text{ét}}(X, \mathbb{G}_m)$ .

Remark 5.10. Note that  $H^2_{\text{\acute{e}t}}(X, \mathbb{G}_m)$  is called "cohomological Brauer group" and denoted Br' X in [48], whereas Br X denotes the "Brauer group" defined in [47] as classes of Azumaya algebras over X.

Using Grothendieck's theorems [47, 48], and in particular the "purity theorem" [39, p. 63, §V, Théorème (3.4); 49, §6], one may show (see [32, §3.4 and Prop. 4.2.3]):

**Theorem 5.11.** Let X be a smooth connected variety over a field k of characteristic zero. Let k(X) be the function field of X.

- (i) There is a natural inclusion  $\operatorname{Br} X \subset \operatorname{Br} k(X)$  in particular  $\operatorname{Br} X$  is a torsion group.
- (ii) The subgroup  $\operatorname{Br}_{\operatorname{nr}}(k(X)/k) \subset \operatorname{Br} k(X)$  lies in  $\operatorname{Br} X$ .
- (iii) If moreover X is proper,  $\operatorname{Br} X = \operatorname{Br}_{\operatorname{nr}}(k(X)/k)$ .

*Remark.* For examples of computations of  $\operatorname{Br}_{\operatorname{nr}}(k(X)/k)$  over a non-algebraically closed field k, see [24, 33, 34].

For later use, let us recall that given any smooth variety X over a field k and any integer n prime to char k, the Kummer sequence in étale cohomology induces a short exact sequence:

$$0 \to \operatorname{Pic} X/n \to H^2_{\operatorname{\acute{e}t}}(X,\mu_n) \to {}_n\operatorname{Br} X \to 0,$$

where Pic X denotes the Picard group of X,  $\mu_n$  the group of *n*-th roots of unity and  $_n \text{Br } X$  the *n*-torsion subgroup of Br X.

### 6. A GENERAL FORMULA

The results of this section are mainly due to F.A. Bogomolov [17, 18]. Here k denotes an algebraically closed field of characteristic zero.

We shall first consider the case of a finite group G.

**Theorem 6.1.** Let G be a finite group of automorphisms of a function field L over the algebraically closed field k of characteristic zero. One then has:

$$\operatorname{Br}_{\operatorname{nr}} L^G = \{ \alpha \in \operatorname{Br} L^G \mid \alpha_H \in \operatorname{Br}_{\operatorname{nr}} L^H \text{ for all } H \in \mathcal{B}_G \},\$$

where  $\mathcal{B}_G$  denotes the set of finite bicyclic subgroups of G (a bicyclic group is an abelian group generated by two elements) and  $\alpha_H$  denotes the restriction of  $\alpha \in \operatorname{Br} L^G$  to  $\operatorname{Br} L^H$ .

*Proof.* Let  $K = L^G$  and let  $\alpha \in \operatorname{Br} K$  be such that  $\partial_A(\alpha) \neq 0$  for some discrete rank one valuation ring  $A \subset K$  with fraction field K. We must show that there exists a subgroup  $H \in \mathcal{B}_G$  such that

$$\alpha_H \notin \operatorname{Br}_{\operatorname{nr}} L^H$$
.

The following facts may be read off from Serre ([136, I, §7]). Let  $\mathfrak{p}$  be a prime ideal in the semi-local Dedekind ring  $\tilde{A}$  which is the integral closure of A in L, and let  $D \subset G$  be the associated decomposition group, and let  $I \subset D$  be the inertia group, which is a normal subgroup of G. The localization  $B = \tilde{A}_{\mathfrak{p}} \subset L$ is a discrete valuation ring. There is a tower of fields:  $K \subset L^D \subset L^I \subset L$ and a corresponding tower of discrete valuation rings obtained by taking the traces  $A = B^G \subset B^D \subset B^I$  of B on the subfields. The corresponding residue field extensions read:  $F = F \subset E = E$ , and we have D/I = Gal(E/F) = $\text{Gal}(L^I/L^D)$ . The Galois extension  $L^I/K$  is unramified, i.e. a uniformizing parameter of A is still a uniformizing parameter in  $B^I$ .

Moreover, since the residue characteristic is zero, the inertia group I may be identified with a cyclic group, namely a group  $\mu$  of roots of unity in F([136, IV, §2, Corollaires 1 and 2]). Furthermore, the conjugacy action of Don the normal subgroup I is then trivial, since this action may be identified with the action of D/I = Gal(E/F) on  $\mu \subset F$ , and all roots of unity are in  $k \subset F$ . Thus I is central in D.

If  $\alpha_I \notin \operatorname{Br}_{\operatorname{nr}} L^I$ , we are done, since I is a cyclic subgroup of G. We may thus assume that  $\alpha_I \in \operatorname{Br}_{\operatorname{nr}}(L^I)$ . Since  $B^D/A$  is an unramified extension of discrete valuation rings which induces an isomorphism on the residue class fields, the assumption  $\partial_A(\alpha) \neq 0$  implies  $\partial_{B^D}(\alpha) \neq 0 \in X(F)$ . On the other hand,  $\partial_{B^I}(\alpha) = 0 \in X(E)$ . Since  $B^I/B^D$  is unramified, the commutative diagram:

$$\begin{array}{ccc} \operatorname{Br} K^{I} & \stackrel{\mathcal{O}_{B^{I}}}{\longrightarrow} & X(E) \\ & \uparrow & & \uparrow^{\operatorname{Res}_{F/E}} \\ \operatorname{Br} K^{D} & \stackrel{\mathcal{O}_{B^{D}}}{\longrightarrow} & X(F) \end{array}$$

implies that  $\partial_{B^D}(\alpha)$  may be identified with a nontrivial character of  $D/I = \operatorname{Gal}(E/F)$ . Let  $g \in D$  be an element of D whose class  $\overline{g}$  in D/I satisfies  $\partial_{B^D}(\alpha)(\overline{g}) \neq 0 \in \mathbb{Q}/\mathbb{Z}$ , let  $H = \langle I, g \rangle \subset D$  be the subgroup spanned by I and g, and let  $F_1$  be the residue class field of  $B^H$ . Inserting  $\operatorname{Br}(K^H) \to X(F_1)$  in the above diagram, one immediately sees that  $\partial(\alpha_H) \neq 0$ , since  $\partial(\alpha_H)$  may be identified with a character of  $\operatorname{Gal}(E/F_1) = D/H$  which does not vanish on  $\overline{g}$ . This is enough to conclude, since H is an extension of the cyclic group  $\langle \overline{g} \rangle$  by the central cyclic subgroup I (see above), hence is an abelian group spanned by two elements.

We now wish to extend Theorem 6.1 to almost free actions of reductive algebraic groups.

**Lemma 6.2.** Let  $p: X \to Y$  be a dominant morphism of smooth integral varieties. If this morphism admits a section over a non-empty open set of Y,

then this section induces a cartesian diagram, where the horizontal maps are injective:



*Proof.* Let the section s be defined over the open set  $V \subset Y$ , and let  $U = p^{-1}(V) \subset X$ . Because of the functorial behaviour of the Brauer group and of the unramified Brauer group we have a commutative diagram:



where the top composite map is identity and where all vertical maps are inclusions.

The right hand part of the diagram requires an explanation, the notation  $s^*$  being a slight abuse of language. Since the characteristic of k is zero, Hironaka's theorem guarantees that the morphism  $p: X \to Y$  extends to a morphism  $p^c: X^c \to Y^c$ , where  $X^c$ , resp.  $Y^c$ , is a smooth, projective integral variety containing X, resp. Y, as a dense open set. Since  $p^c$  is a proper morphism and  $Y^c$  is smooth, the section  $s: V \to X$  extends to a section  $s^c: W \to X^c$  of  $p^c$ , where  $W \subset Y^c$  is an open set which contains all codimension 1 points of  $Y^c$ :



By purity of the Brauer group (Theorem 5.11), the restriction map  $\operatorname{Br} Y^c \to \operatorname{Br} W$  is an isomorphism, and both groups coincide with  $\operatorname{Br}_{\operatorname{nr}} k(Y)$ . We thus have the commutative diagram:

$$\begin{array}{c} \operatorname{Br} U \xrightarrow{s^*} \operatorname{Br} V \\ & & & \\ & & \\ \operatorname{Br} X \xrightarrow{s^*} \operatorname{Br} V \\ & & \\ & & \\ \operatorname{Br}_{\operatorname{nr}} k(X) = \operatorname{Br} X^{\operatorname{c}} \xrightarrow{s^{\operatorname{c}*}} \operatorname{Br} W \xleftarrow{\cong} \operatorname{Br} Y^{\operatorname{c}} = \operatorname{Br}_{\operatorname{nr}} k(Y). \end{array}$$

The composite map  $\operatorname{Br} X^{\operatorname{c}} \to \operatorname{Br} X \to \operatorname{Br} U \to \operatorname{Br} V$  coincides with the composite map  $\operatorname{Br} X^{\operatorname{c}} \to \operatorname{Br} W \to \operatorname{Br} V$ , which may be rewritten  $\operatorname{Br}_{\operatorname{nr}} k(X) \to \operatorname{Br}_{\operatorname{nr}} k(Y) \to \operatorname{Br} V$ .

Now if  $\alpha \in \operatorname{Br} Y$  is such that  $p^*(\alpha) \in \operatorname{Br} X$  actually lies in  $\operatorname{Br}_{\operatorname{nr}} k(X)$ , the above diagram shows that  $\operatorname{Res}_{Y/V}(\alpha) = s^* p^*(\operatorname{Res}_{Y/V}(\alpha))$  belongs to  $\operatorname{Br}_{\operatorname{nr}} k(Y)$ , hence also  $\alpha \in \operatorname{Br}_{\operatorname{nr}} k(Y)$ .

**Lemma 6.3** (Bogomolov [17]). Let k be an algebraically closed field, char k = 0, and let G be a reductive algebraic group over k which is an extension of a finite group W by a torus T. Let X be an integral affine k-variety with an action of G. Assume that all stabilizers are trivial. Then, there exists a finite subgroup A of G such that the natural map  $X/A \to X/G$  has a section over a non-empty open set of X/G.

*Proof.* First af all, for any algebraic subgroup H of G, the natural map  $X \to X/\!\!/H$  makes X into a H-torsor over  $X/\!\!/H$ . Hence X/H exists, and  $X/H = X/\!\!/H$ . By assumption, the group G defines an extension:

$$1 \to T \to G \to W \to 1$$
,

hence an algebraic action of W on T, and a class in  $H^2(W,T)$ . Let  ${}_nT$  be the subgroup of *n*-torsion points of T for  $n \ge 1$ , and let  $T_{\text{tors}} = \bigcup_{n\ge 1} {}_nT$  be the whole torsion subgroup of T. The exact sequence

$$1 \to T_{\text{tors}} \to T \to T \otimes_{\mathbb{Z}} \mathbb{Q} \to 1$$

and the vanishing of  $H^i(W, T \otimes_{\mathbb{Z}} \mathbb{Q})$  for  $i \geq 1$  shows that this class comes from a (unique) class in  $H^2(W, T_{\text{tors}})$ , hence from some class in  $H^2(W, nT)$  for some n. Thus there is a finite group  $H_n$  and a commutative diagram of extensions:

$$1 \longrightarrow T \longrightarrow G \longrightarrow W \longrightarrow 1$$

$$\uparrow \qquad \uparrow \qquad \parallel$$

$$1 \longrightarrow {}_{n}T \longrightarrow H_{n} \longrightarrow W \longrightarrow 1.$$

This diagram gives rise to the following fibre product:

$$\begin{array}{cccc} X/_nT & \longrightarrow & X/H_n \\ q_n \downarrow & & \downarrow p_n \\ X/T & \longrightarrow & X/G. \end{array}$$

Moreover, the map  $q_n$  makes  $X/_nT$  into a torsor over X/T under the torus  $T = T/_nT$ . Note that the group W acts upon  $X/_nT$  and X/T (indeed, both horizontal maps in the above diagram are Galois coverings with group W). The map  $q_n$  is W-equivariant. Since torsors under tori are locally trivial for the Zariski topology, the set of rational sections of  $q_n$  is not empty. The group W acts upon this set.

If we may find such a section which is W-invariant (for the obvious action induced by the action of W on both spaces  $X/_nT$  and X/T), then this section descends to a section of  $p_n$  and we get the conclusion of the lemma.

We shall show that at the cost of changing n into nm for a suitable  $m \ge 1$ , there exists such a section. Let  $E = k(X)^T = k(X/T)$  and let  $F = k(X)^G = k(X/G)$ . The extension E/F is a Galois extension with Galois group W. Thus to the action of W on the k-torus T we may associate a unique (twisted) Ftorus R which becomes isomorphic to the torus  $T_E = T \times_k E$  over E. Namely, we take

$$R = \operatorname{Spec} F[X(T)]^W,$$

where F[X(T)] is the group algebra over the character group X(T) of T (which is a free abelian group of finite type), and the W-action on F[X(T)] is simultaneous on F and X(T). The generic fibre of  $q_n$  is a principal homogenous space under the action of  $T_E$ . Since W acts equivariantly on the whole situation, the generic fibre  $P_n$  of  $p_n$ , which is an F-variety, inherits a structure of principal homogeneous space under the F-torus R. The isomorphy class of this principal homogeneous space in the étale (= Galois) cohomology group  $H^1(F, R)$  is killed by some integer  $m \ge 1$ . Let us then consider the commutative diagram of exact sequences:

which reflects the *W*-action on the commutative diagram

We also have the fibre product:

$$\begin{array}{cccc} X/_{mn}T & \longrightarrow & X/H_{mn} \\ & & & & \downarrow p_{mn} \\ & & & & \downarrow p_{mn} \\ & & & X/T & \longrightarrow & X/G, \end{array}$$

where  $H_{nm}$  is the finite group obtained from  $H_n$  by pushing out through  ${}_{n}T \rightarrow {}_{mn}T$ , thus yielding the commutative diagram:

Both the generic fibre  $P_n$  of  $q_n$  and  $P_{nm}$  of  $q_{nm}$  are principal homogeneous spaces under the same *F*-torus *R*. Indeed, on  $X/_nT$ , the torus which acts is the quotient  $T = T/_nT$ , and on  $X/_nT$ , it is  $T = T/_nT$ . Now as the

previous diagrams reveal,  $r: X/_nT \to X/_{nm}T$  is the obvious projection, t is in T and x is in  $X/_nT$ , then  $r(t.x) = t^m \cdot r(x)$ . Thus the natural map from  $X/H_n \to X/H_{nm}$  induces on generic fibres a map  $r_1: P_n \to P_{nm}$  which satisfies

$$r_1(t \cdot x) = t^m \cdot r_1(x)$$
 for  $t \in R$ .

This implies that the class of  $P_{nm}$  in  $H^1(F, R)$  is m times the class of  $P_n$ . Since the class of  $P_n$  is killed by m, we conclude that the class of  $P_{nm}$  is trivial, which completes the proof.

**Theorem 6.4** (Bogomolov [18, Theorem 2.1]). Let k be an algebraically closed field of characteristic zero, let G be a reductive group over k, and let X be an integral affine k-variety with a G-action. Assume that all stabilizers are trivial. One then has

$$\operatorname{Br}_{\operatorname{nr}} k(X)^G = \{ \alpha \in \operatorname{Br} k(X)^G \mid \alpha_A \in \operatorname{Br}_{\operatorname{nr}} k(X)^A \text{ for all } A \in \mathcal{B}_G \},\$$

where  $\mathcal{B}_G$  denotes the set of finite bicyclic subgroups of G(k) and  $\alpha_A$  denotes the restriction of  $\alpha \in \operatorname{Br} k(X)^G$  to  $\operatorname{Br} k(X)^A$ .

*Proof.* Let  $G^{\circ}$  be the connected component of identity, and let  $\mathfrak{g}$  be its Lie algebra. Let the group G act on  $\mathfrak{g}$  via the adjoint representation and let G act upon the product  $X \times \mathfrak{g}$  via the diagonal action. Let  $\mathfrak{t} \subset \mathfrak{g}$  denote a fixed Cartan subalgebra, and let  $N \subset G$  be the normalizer of t for the adjoint action. The group N is a reductive group whose identity component is a (maximal) torus  $T \subset G$ . It is a classical fact that any regular semisimple element in  $\mathfrak{g}$  is conjugate under G to an element of  $\mathfrak{t}$ . Moreover, on the open set  $\mathfrak{t}^{\circ} \subset \mathfrak{t}$  consisting of regular semisimple elements, the following property holds: if  $g \in G$ ,  $x, y \in \mathfrak{t}^{\circ}$  and  $g \cdot x = y$ , then  $g \in N$ . Thus the closed subvariety  $X \times \mathfrak{t} \subset X \times \mathfrak{g}$  is a (G, N)-slice, and by the slice lemma 3.1 we have  $k(X \times \mathfrak{t})^N \cong k(X \times \mathfrak{g})^G$ . On the other hand, the no-name lemma 3.9 implies that  $k(X \times \mathfrak{t})^N$ , resp.  $k(X \times \mathfrak{g})^G$ , is purely transcendental over  $k(X)^N$ , resp.  $k(X)^G$ . According to Proposition 5.7, we thus have:  $\operatorname{Br}_{\operatorname{nr}} k(X)^N \cong \operatorname{Br}_{\operatorname{nr}} k(X \times X)^N$  $\mathfrak{t}^{N}$  and  $\operatorname{Br}_{\operatorname{nr}} k(X)^{G} \cong \operatorname{Br}_{\operatorname{nr}} k(X \times \mathfrak{g})^{G}$ . The functoriality of the unramified Brauer group now implies that the natural inclusion  $k(X)^{G} \subset k(X)^{N}$  induces an isomorphism  $\operatorname{Br}_{\operatorname{nr}} k(X)^N \cong \operatorname{Br}_{\operatorname{nr}} k(X)^G$ . According to Lemma 6.3, there exists a finite group  $H \subset N$  such that the map between quotient spaces:  $X/H \rightarrow X/N$  has a rational section. Now Lemma 6.2 shows that an element  $\alpha \in Br(k(X)^N)$  is unramified if and only if its restriction  $\alpha_H$  is unramified. By Theorem 6.1,  $\alpha_H$  itself is unramified if and only if  $\alpha_A$  is unramified for all bicyclic subgroups A of H. The theorem follows.  $\square$ 

#### 7. LINEAR ACTION OF A FINITE GROUP

In this section, k denotes an algebraically closed field of characteristic zero.

**Theorem 7.1** (Bogomolov [17]). Let  $G \subset GL(V)$  be a finite group of automorphisms of a finite dimensional k-vector space V. One then has:

$$\begin{aligned} \operatorname{Br}_{\operatorname{nr}} k(V)^{G} &\cong \operatorname{ker} \left[ H^{2}(G, k^{*}) \xrightarrow{\operatorname{Res}} \prod_{A \in \mathcal{B}_{G}} H^{2}(A, k^{*}) \right] \\ &\cong \operatorname{ker} \left[ H^{2}(G, \mathbb{Q}/\mathbb{Z}) \xrightarrow{\operatorname{Res}} \prod_{A \in \mathcal{B}_{G}} H^{2}(A, \mathbb{Q}/\mathbb{Z}) \right] \\ &\cong \operatorname{ker} \left[ H^{3}(G, \mathbb{Z}) \xrightarrow{\operatorname{Res}} \prod_{A \in \mathcal{B}_{G}} H^{3}(A, \mathbb{Z}) \right], \end{aligned}$$

where  $\mathcal{B}_G$  denotes the set of all bicyclic subgroups of G (in the last two formulas, G acts trivially upon  $\mathbb{Z}$  and  $\mathbb{Q}/\mathbb{Z}$ ).

The same formulas hold if one replaces the set  $\mathcal{B}_G$  of bicyclic subgroups by the set  $\mathcal{A}_G$  of all abelian subgroups of G.

*Proof.* If  $A \subset G$  is any abelian group, the field  $k(V)^A$  is pure over k by Fischer's theorem (see Proposition 4.3), hence  $\operatorname{Br}_{\operatorname{nr}} k(V)^A = 0$ . On the other hand, given any subgroup  $H \subset G$ , the restriction-inflation sequence for the faithful action of H on  $k(V)^*$  yields an exact sequence:

$$0 \to H^2(H, k(V)^*) \to \operatorname{Br} k(V)^H \to \operatorname{Br} k(V).$$

Using the functoriality of the unramified Brauer group 5.5 and Theorem 6.1, one gets the formula:

$$\operatorname{Br}_{\operatorname{nr}} k(V)^G \cong \operatorname{ker} \Big[ H^2(G, k(V)^*) \xrightarrow{\operatorname{Res}} \prod_{A \in \mathcal{B}_G} H^2(A, k(V)^*) \Big].$$
(7.1)

Since k[V] is a UFD and  $k[V]^* = k^*$ , one has a short exact sequence of *G*-modules:

$$1 \to k^* \to k(V)^* \to \operatorname{Div} V \to 0,$$

where the G-module Div V is a direct sum of permutation modules  $\mathbb{Z}[G/H]$  for various subgroups H. Such a module satisfies the two properties:

$$\begin{split} H^1(G,M) &= 0, \\ & \ker \Big[ H^2(G,M) \to \prod_{g \in G} H^2(\langle g \rangle, M) \Big] = 0, \end{split}$$

as one easily checks by reducing to the case  $M = \mathbb{Z}$  with trivial action. The first formula now easily follows. The second formula is obtained by identifying the group  $\mu$  of roots of unity in  $k^*$  with  $\mathbb{Q}/\mathbb{Z}$ , and using the unique divisibility, hence cohomological triviality of  $k^*/\mu$ . As for the third formula, it is obtained by shifting via the exact sequence  $0 \to \mathbb{Z} \to \mathbb{Q} \to \mathbb{Q}/\mathbb{Z} \to 0$ .

The last statement now follows from the vanishing  $\operatorname{Br}_{\operatorname{nr}} k(V)^A = 0$  for any abelian subgroup A of G.

7.1. The case of a nilpotent group of class 2. Bogomolov [17] made a thorough application of the previous theorem when G is a nilpotent group, particularly of class 2. Such a group is a central extension

$$1 \to C \to G \to \Gamma \to 1$$

of a finite abelian group  $\Gamma$  by another finite abelian group C and classes of such extensions are classified by the group  $H^2(\Gamma, C)$  where C is viewed as a trivial  $\Gamma$ -module. We denote by  $[G] \in H^2(\Gamma, C)$  the class of G.

It is well known [26, §V.6, Theorem 6.4] that for an abelian group  $\Gamma$  we have  $H_2(\Gamma, \mathbb{Z}) = \Lambda^2 \Gamma$ . Then, for any  $\Gamma$ -module M, the universal coefficient sequence [26, §V.6, exercise 5] yields

$$0 \to \operatorname{Ext}^1(\Gamma, M) \to H^2(\Gamma, M) \xrightarrow{\omega_M} \operatorname{Hom}(\Lambda^2\Gamma, M) \to 0.$$

Then for  $M = \mathbb{Q}/\mathbb{Z}$  with trivial action, the map

$$\omega_{\mathbb{Q}/\mathbb{Z}} \colon H^2(\Gamma, \mathbb{Q}/\mathbb{Z}) \to \operatorname{Hom}(\Lambda^2\Gamma, \mathbb{Q}/\mathbb{Z})$$

is an isomorphism. For M = C with trivial action, the map

$$\omega_C \colon H^2(\Gamma, C) \to \operatorname{Hom}(\Lambda^2 \Gamma, C)$$

sends the class of [G] to

$$\lambda_G \colon \Lambda^2 \Gamma \to [G, G] \subset C$$

defined by

$$\lambda_G(\gamma_1 \land \gamma_2) = [g_1, g_2]$$

where  $g_1, g_2 \in G$  are lifts of  $\gamma_1, \gamma_2 \in \Gamma$ . From the definition  $\text{Im } \lambda_G = [G, G]$ . So  $\lambda_G$  is surjective if and only if [G, G] = C.

For any subgroup G' of G, we denote by  $S_{G'}$  the kernel of  $\lambda_{G'} \colon \Lambda^2 \Gamma' \to C'$ , where  $\Gamma'$  is the image of G' in  $\Gamma$  and  $C' = G' \cap C$ .

**Definition 7.2.** Let G be a central extension of a finite abelian group  $\Gamma$  by another one C. To the canonical homomorphism  $\lambda_G \colon \Lambda^2 \Gamma \to C$  one may attach the following subgroups

$$S_{
m bic} = S_{G,
m bic} \subset S_G := \ker \lambda_G \subset \Lambda^2 \Gamma$$

where  $S_{\text{bic}}$  is the subgroup of  $S_G$  generated by all the  $\gamma_1 \wedge \gamma_2$  which belong to  $S_G$ .

In other words,  $S_G$  is defined by the exact sequence

$$0 \to S_G \to \Lambda^2 \Gamma \xrightarrow{\lambda_G} C$$

and  $S_{\text{bic}}$  is the subgroup generated by the images of the  $S_A$ 's for all bicyclic subgroups A of G.

**Theorem 7.3** (Bogomolov [17]). Let G be a central extension of a finite abelian group  $\Gamma$  by another one C. Then

$$\operatorname{Br}_{\operatorname{nr}} k(V)^G = \operatorname{ker}(\widehat{S_G} \to \widehat{S_{\operatorname{bic}}}) = (\widehat{S_G}/\widehat{S_{\operatorname{bic}}})$$

where  $S_G$  is the kernel of the canonical morphism  $\lambda_G \colon \Lambda^2 \Gamma \to C$ , and

$$S_{\rm bic} \subset S_G \subset \Lambda^2 \Gamma$$

is the subgroup generated by the  $S_A$ 's for all bicyclic subgroups A of G.

*Proof.* The lower terms exact sequence of the Hochschild-Serre spectral sequence

$$H^p(\Gamma, H^q(C, \mathbb{Q}/\mathbb{Z})) \implies H^{p+q}(G, \mathbb{Q}/\mathbb{Z})$$

yields the top exact sequence in the following diagram:

$$\widehat{C} \longrightarrow H^2(\Gamma, \mathbb{Q}/\mathbb{Z}) \longrightarrow \ker[H^2(G, \mathbb{Q}/\mathbb{Z}) \to H^2(C, \mathbb{Q}/\mathbb{Z})] \longrightarrow H^1(\Gamma, \widehat{C})$$

$$\left\| \begin{array}{c} \omega_{\mathbb{Q}/\mathbb{Z}} \\ \widehat{C} \end{array} \right\| \longrightarrow \operatorname{Hom}(\Lambda^2\Gamma, \mathbb{Q}/\mathbb{Z}) \longrightarrow \operatorname{Hom}(S_G, \mathbb{Q}/\mathbb{Z}) = \widehat{S_G} \longrightarrow 0.$$

This gives the following exact sequence:

$$0 \to \widehat{S_G} \to \ker \left[ H^2(G, \mathbb{Q}/\mathbb{Z}) \to H^2(C, \mathbb{Q}/\mathbb{Z}) \right] \to H^1(\Gamma, \widehat{C}).$$

We want to calculate

$$B_G := \operatorname{Br}_{\operatorname{nr}} \mathbb{C}(V)^G \subset H^2(G, \mathbb{Q}/\mathbb{Z}).$$

This inclusion is functorial and covariant in G. Fischer's theorem yields  $B_A = 0$  for any abelian group A. In particular,  $B_C = 0$ . So by restriction  $B_G$  maps to 0 in  $H^2(C, \mathbb{Q}/\mathbb{Z})$ :

$$B_G \subset \ker \left[ H^2(G, \mathbb{Q}/\mathbb{Z}) \to H^2(C, \mathbb{Q}/\mathbb{Z}) \right].$$

Then we prove that  $B_G$  maps to 0 in  $H^1(\Gamma, \widehat{C}) = \text{Hom}(\Gamma, \widehat{C})$ . By functoriality, it is sufficient to prove this result by restriction to any cyclic subgroup  $\Gamma'$  of  $\Gamma$ . Let G' the restriction of the extension G to  $\Gamma'$ . A central extension of a cyclic group is an abelian group. Thus  $\Gamma'$  is abelian and again by Fisher's theorem  $B_{G'} = 0$ .

We finally get the inclusion

$$B_G \subset \widehat{S_G}.$$

Theorem 7.1 says that

$$B_G \cong \ker \left[ H^2(G, \mathbb{Q}/\mathbb{Z}) \to \prod_{G' \in \mathcal{B}_G} H^2(G', \mathbb{Q}/\mathbb{Z}) \right]$$

where G' is any bicyclic subgroup of G. Then by functoriality  $B_G$  is the subgroup of  $\widehat{S}_G$  which maps to 0 in each of the  $H^2(G', \mathbb{Q}/\mathbb{Z})$ 's, hence to 0 in each of the  $\widehat{S}_{G'}$ 's for any bicyclic subgroup G' of G. Let  $\Gamma'$  be the image in  $\Gamma$  of such a subgroup G'. We have a commutative diagram

Let  $g_1, g_2 \in G$  and  $\gamma_1, \gamma_2 \in \Gamma$  their images. Then the two conditions

- (i)  $g_1g_2 = g_2g_1$ ,
- (ii)  $\gamma_1 \wedge \gamma_2 \in S_G$ ,

are equivalent.

Hence the cyclic subgroups of  $S_G$  generated by elements of the form  $\gamma_1 \wedge \gamma_2$  are exactly the images of the  $S_{G'}$ 's for G' a bicyclic subgroup of G.

**Corollary 7.4.** Let  $\Gamma$  be a finite abelian group. There is a bijection between the classes of central extensions G of  $\Gamma$  such that  $B_G \neq 0$  and the subgroups

 $S \subset \Lambda^2 \Gamma$ 

such that

 $S_{\rm bic} \neq S.$ 

Examples 7.5. Let us find simple examples of finite groups G such that  $B_G \neq 0$ . We consider only nilpotent groups G of class 2.

We are searching G among central extensions of a fixed finite abelian group  $\Gamma$  by using Corollary 7.4.

Consider the case  $\Gamma = (\mathbb{Z}/p)^4$ . We find  $\Lambda^2 \Gamma = \mathbb{F}_p^6$ . It is simpler to work in the projective space  $\mathbb{P}(\Lambda^2 \Gamma) = \mathbb{P}^5(\mathbb{F}_p)$ . For  $S \subset \Lambda^2 \Gamma = \mathbb{F}_p^6$  the following conditions are equivalent:

- (i)  $S \neq S_{\text{bic}}$ .
- (ii)  $\mathbb{P}(S) \cap Q$  does not generate  $\mathbb{P}(S)$  as a linear space, where Q denotes the quadric whose equation is  $x_1x_2 + x_3x_4 + x_5x_6 = 0$  which consists of all undecomposable tensors  $\gamma_1 \wedge \gamma_2$ .

Here is the list of all possibilities for condition (ii):

- 1.  $\mathbb{P}(S) =$  a point  $\notin Q$  (this is Saltman's original example [119]; here  $\#G = p^9$ ).
- 2.  $\mathbb{P}(S) = a$  line tangent to Q (here  $\#G = p^8$ ).
- 3.  $\mathbb{P}(S) = a$  line which does not intersect Q (on  $\mathbb{F}_p$ ) (here  $\#G = p^8$ ).
- 4.  $\mathbb{P}(S) =$  a 2-plane which intersects Q along a line (here  $\#G = p^7$ ).
- 5.  $\mathbb{P}(S) = a$  2-plane which intersects Q in a point (here  $\#G = p^7$ ).

More generally, for  $\Gamma = (\mathbb{Z}/p)^{2m}$ , the subvariety of  $\mathbb{P}(\Lambda^2\Gamma)$  which consists of tensors in  $\Lambda^2\Gamma$  of rank < 2m is a hypersurface of degree m which is the zero locus of the Pfaffian. Among different examples, the simplest one is obtained by taking  $\mathbb{P}(S)$  to be a point outside this hypersurface. One may also consider the subspace S which consists of all matrices

$$\begin{pmatrix} M_{1,1} & M_{1,2} \\ M_{2,1} & M_{2,2} \end{pmatrix}$$

whose 4 blocks are  $m \times m$  matrices of the following type:

$$M_{1,1} = M_{2,2} = 0$$

 $M_{1,2}$  trigonal with diagonal entries  $\{\lambda, \ldots, \lambda\}$ .

in which case  $S_{\text{bic}}$  is the subspace  $\lambda = 0$ .

*Remark.* In [17] Bogomolov also gives an example of a group G of order  $p^6$  with  $B_G \neq 0$  and proves that  $B_G = 0$  for any G of order  $p^n$  for  $n \leq 5$ . As a matter of fact, for  $n \leq 4$ , the fields  $k(V)^G$  are stably rational (Chu and Kang [30]). The stable rationality for n = 5 remains open. Recent computations of  $B_G$  for finite Chevalley groups of type  $A_n$  may be found in [21].

#### 8. Multiplicative action of a finite group

Most results in the present section are due to D. Saltman. Here k denotes an algebraically closed field of characteristic zero.

Let G be a group. A G-lattice M is a Z-free module M of finite rank equipped with a G-action. We denote k[M] the group algebra of M over k. If M is of rank r, then  $k[M] \cong k[t_1, t_1^{-1}, \ldots, t_r, t_r^{-1}]$ . The associated field of fractions is denoted  $k(M) \cong k(t_1, \ldots, t_r)$ .

**Definition 8.1.** Given a *G*-lattice *M* consider the action of *G* on k[M] which is trivial on *k* and coincides with the given one on *M*. The induced action of *G* on k(M) is called the *multiplicative action* of *G* associated to the *G*-lattice *M*.

Such actions are sometimes referred to as "purely monomial actions".

**Definition 8.2.** For a given G-lattice M we may also consider the *twisted* multiplicative action of G associated to a given crossed homomorphism of the finite group G in  $\widehat{M} = \text{Hom}(M, k^*)$ , i.e. to a 1-cocycle  $\alpha \in Z^1(G, \widehat{M})$ . If we denote by  $t^m$  the canonical image of  $m \in M$  in k[M] the  $\alpha$ -twisted action of G on the k-algebra k[M] is given by:

$$g \cdot_{\alpha} t^m = \alpha_{q^{-1}}(m) t^{g \cdot m}.$$

with trivial action on the coefficients. We denote by  $k[M]_{\alpha}$  this twisted Gk-algebra and by  $k(M)_{\alpha}$  the field of fractions. Let  $\alpha$  and  $\alpha'$  be in the same cohomology class. Then the two G-k-algebras  $k[M]_{\alpha}$  and  $k[M]_{\alpha'}$  are isomorphic: if  $\alpha'/\alpha = d\beta$  where  $\beta : M \to k^*$  then  $t^m \mapsto \beta(m)t^m$  defines a morphism of k-algebras " $\beta$ ":  $k[M]_{\alpha} \to k[M]_{\alpha'}$  which is G-equivariant and which is clearly an isomorphism. Note that the formula

$$g \cdot_{\alpha} t^m = \alpha_a (g \cdot m) t^{g \cdot m}$$

defines the opposite twisted action on k[M], since we have

$$\alpha_g(g \cdot m) = ({}^{g^{-1}}\alpha_g)(m) \text{ and } 1 = \alpha_e = \alpha_{g^{-1}}({}^{g^{-1}}\alpha_g).$$

A good reason for studying such actions is provided by the following theorem, which extends early work of Procesi [109] and Formanek [43] for  $G = PGL_n$ :

**Theorem 8.3** (Saltman [124]). Let G be a reductive connected linear algebraic group over k. Let V be a finite-dimensional k-vector space with an almost free linear G-action. Let  $T \subset G$  be a maximal torus,  $W = N_G(T)/T$  the Weyl group and X(T) the character group of T, viewed as a W-lattice. Choose a surjective map  $f: P \to X(T)$  with P a permutation W-lattice. Let M be the kernel of f. Then  $k(V)^G$  is stably isomorphic over k to  $k_{\alpha}(M)^W$  for suitable  $\alpha \in Z^1(W, \operatorname{Hom}(M, k^*))$ .

For further work in this direction, see [9, 11, 132, 133].

**Lemma 8.4.** Every twisted G-k-algebra  $k[M]_{\alpha}$  defines an extension of Gmodules

$$1 \to k^* \to k[M]^*_{\alpha} \to M \to 0$$

whose class is precisely the class of  $\alpha \in Z^1(G, \operatorname{Hom}(M, k^*))$ .

**Lemma 8.5.** Any twisted multiplicative action of a group G on a permutation G-lattice P extends to a G-linear action on a vector space V such that  $k(M)^G = k(V)^G.$ 

*Proof.* Let  $\{e_x\}$  a basis of the  $\mathbb{Z}$ -lattice P which is G-stable. Then G acts linearly on  $V = \bigoplus_{x} ke_{x}$ . As k[P] is obtained from k[V] by inverting the x's we have the natural G-equivariant inclusions

$$k[V] \subset k[P] \subset k(V) = k(P)$$

and the same fields of invariants  $k(P)^G = k(V)^G$ .

**Lemma 8.6.** Given a multiplicative action of a group G on a lattice M one may find a G-lattice N, a finite group  $\Gamma$  which is a semi-direct extension of G by an abelian group and a faithful linear action of  $\Gamma$  on a finite dimensional vector space V such that  $k(M \oplus N)^G = k(V)^{\Gamma}$ .

*Proof.* Up to changing M by  $M \oplus N$ , where N is an auxiliary G-lattice, one may find an exact sequence of G-modules

$$0 \to M \to P \xrightarrow{\chi} F \to 0,$$

where P is a permutation G-lattice and F is finite. The extension k(P)/k(M)is Galois with group  $\widehat{F} = \operatorname{Hom}(F,\mu)$  (where  $\mu \subset k^*$  denotes the group of roots of unity in k). The G-action on k(M) extends to k(P). Then the extension  $k(P)/k(M)^G$  is also Galois with Galois group  $\Gamma$  which is the semidirect product of  $\widehat{F}$  with G:

$$\Gamma = \widehat{F} \rtimes G.$$

The group  $\Gamma$  acts linearly on k(P). This is clear for  $g \in G$ . One can verify that each  $\gamma \in \widehat{F}$  acts on  $P \subset k(P)$  through multiplication by roots of unity:

$$\gamma \cdot p = \langle \gamma, \chi(p) \rangle p \text{ for } \gamma \in F \text{ and } p \in P.$$

Thus  $\Gamma$  acts linearly on  $k \otimes_{\mathbb{Z}} P$ .

For  $M = \mathbb{Z}[G]$  the following theorem reduces to Theorem 7.1.

**Theorem 8.7** (Saltman [126, Theorem 12]). Let G be a finite group, let Mbe a faithful G-lattice, and let k(M) denote the field of fractions of the group algebra k[M]. One then has:

$$\operatorname{Br}_{\operatorname{nr}} k(M)^G \cong \ker \Big[ H^2(G, k^* \oplus M) \xrightarrow{\operatorname{Res}} \prod_{A \in \mathcal{B}_G} H^2(A, k^* \oplus M) \Big].$$

*Proof.* The variety  $X = \operatorname{Spec} k[M]$  is isomorphic to a product of copies of the multiplicative group  $\mathbb{G}_m$ . Hence it is factorial (Pic X = 0). Note that Div X is always a direct sum of permutation modules. Moreover, the group of units

 $k[M]^*$  may be identified with the group  $k^*\oplus M$  (as G-modules). Thus one has an exact sequence of G-modules:

$$1 \to k^* \oplus M \to k(M)^* \to \operatorname{Div} X \to 0.$$

If we argue as in Theorem 7.1, the theorem will follow from the following propositions:

**Proposition 8.8.** Let A be a cyclic group, and let M be an A-lattice. Then the field of invariants  $k(M)^A$  is retract rational and  $\operatorname{Br}_{nr} k(M)^A = 0$ .

*Proof.* Replacing A by its image in Aut M allows us to assume that M is a faithful A-module. Let  $0 \to M \to P \to F \to 0$  be a flasque resolution of M. Here P is a permutation G-lattice and F is a flasque A-lattice, i.e.  $H^1(H, \operatorname{Hom}(F,\mathbb{Z})) = 0$  for all subgroups H of A. Such resolutions always exist [36]. If A is cyclic, a basic result of Endo and Miyata says that any flasque A-lattice is a direct factor of a permutation A-lattice, see [36]. Quite generally, given such an exact sequence as above, the field  $k(P)^A/k(M)^A$  is the function field of a principal homogeneous space E over  $k(M)^A$  under the  $k(M)^A$ -torus whose character group over k(M) is the A-lattice F. Since F here is a direct factor of a permutation module, it follows from Hilbert's theorem 90 that any such principal homogeneous space is trivial. In particular, E has a  $k(M)^A$ -rational point, and by Lemma 6.2 Br<sub>nr</sub>  $k(M)^A$  injects into Br<sub>nr</sub>  $k(P)^A$ . This last group is trivial, because A is abelian and P is a permutation module, hence  $k(P)^A$  is pure over k by Fischer's theorem. □

**Proposition 8.9.** Let A be a bicyclic group, and let M be an A-lattice. Then  $\operatorname{Br}_{\operatorname{nr}} k(M)^A = 0.$ 

*Proof.* The proof of this proposition is much more technical. Indeed it is the core of the proof of Theorem 8.7. We shall refer to Saltman's original paper [126] and to Barge's more natural proof [4].  $\Box$ 

Remark 8.10. That the kernel ker $[H^2(G, k^* \oplus M) \xrightarrow{\text{Res}} \prod_{A \in \mathcal{B}_G} H^2(A, k^* \oplus M)]$ is a subgroup of  $\text{Br}_{nr} k(M)^G$  does not rely on the last two propositions. This accounts for Saltman's early counter-examples to the "Noether problem" over an algebraically closed field [119, 122].

**Proposition 8.11** (Saltman [122]). Let G be a finite group of order  $p^n$ , p prime, and assume that G is neither cyclic nor bicyclic – which implies  $n \ge 3$ . Let M be a faithful G-lattice and assume  $\exp H^2(G, M) = p^n$ . Then  $\operatorname{Br}_{nr} k(M)^G \neq 0$ .

*Proof.* By hypothesis there exists an element  $\alpha \in H^2(G, M)$  of order  $p^n$ . Then  $0 \neq \beta := p^{n-1} \alpha \in H^2(G, M).$ 

If A is a cyclic or a bicyclic subgroup subgroup of G, then  $A \neq G$  hence  $\#A = p^m$  with  $m \leq n-1$ . In particular, the group  $H^2(A, M)$  is killed by  $p^{n-1}$  and the restriction of  $\beta = p^{n-1}\alpha$  to  $H^2(A, M)$  is trivial. Hence

$$0 \neq \beta \in \ker \left[ H^2(G, M) \xrightarrow{\operatorname{Res}} \prod_{A \in \mathcal{B}_G} H^2(A, M) \right] \subset \operatorname{Br}_{\operatorname{nr}} k(M)^G$$

where the last inclusion is part of Theorem 8.7.

Example 8.12 (multiplicative example [122]). Let G be as above, i.e. of order  $\#G = p^n$ , p prime and G not bicyclic which implies  $n \ge 3$ . In the standard resolution of the trivial G-lattice  $\mathbb{Z}$ , let M denote the kernel of the map  $\pi \colon \mathbb{Z}[G \times G] \to \mathbb{Z}[G]$  given on generators of  $\mathbb{Z}[G \times G]$  by  $(g,h) \mapsto (g-h)$ . From the exact sequence

$$0 \to M \to \mathbb{Z}[G \times G] \xrightarrow{\pi} \mathbb{Z}[G] \to \mathbb{Z} \to 0$$

we deduce  $H^2(G, M) \cong \widehat{H}^0(G, \mathbb{Z}) \cong \mathbb{Z}/p^n$  where  $\#G = p^n$ . Then

$$\exp H^2(G, M) = p^n.$$

This gives for every such group G a faithful G-lattice M such that

$$\operatorname{Br}_{\operatorname{nr}} k(M)^G \neq 0.$$

For a given prime p the simplest group is  $G = (\mathbb{Z}/p)^3$ .

*Example* 8.13 (linear example). Using Lemma 8.6, for any prime p this example leads to a linear example for a group  $\Gamma$  of order  $p^{3p^3}$ .

**Corollary 8.14** (Barge [4]). For a finite group G, the following conditions are equivalent:

- (i) For every faithful G-lattice M, we have  $\operatorname{Br}_{\operatorname{nr}} k(M)^G = 0$ .
- (ii) All Sylow subgroups of G are bicyclic.

*Proof.* Let us prove that (ii)  $\implies$  (i). Let p be a prime and  $G_p \subset G$  a p-Sylow subgroup. The field extension  $k(M)^{G_p}/k(M)^G$  is of degree prime to p. It therefore induces an imbedding on p-primary components of Brauer groups, and also of unramified Brauer groups:

$$\operatorname{Br}_{\operatorname{nr}}(k(M)^G)(p \subset \operatorname{Br}_{\operatorname{nr}}(k(M)^{G_p})(p,$$

and the later group vanishes according to Proposition 8.9 applied to the bicyclic group  $G_p$ .

For the converse, assume there exists an  $\ell$ -Sylow subgroup  $G_{\ell} \subset G$  which is not bicyclic. Let  $\ell^n$  be its order. Let M be the faithful G-lattice defined by the exact sequence from Example 8.12:

$$0 \to M \to \mathbb{Z}[G \times G] \xrightarrow{\pi} \mathbb{Z}[G] \to \mathbb{Z} \to 0.$$

Then

$$H^2(G,M) \cong \bigoplus_p \mathbb{Z}/p^{n_p}$$

where  $\prod_p p^{n_p}$  is the order of the group G. Let  $\beta \in H^2(G, M)$  be the element defined in  $\bigoplus_p \mathbb{Z}/p^{n_p}$  by

$$\beta_p = \begin{cases} \ell^{n-1} & \text{if } p = \ell, \\ 0 & \text{if } p \neq \ell. \end{cases}$$

The same argument as in Proposition 8.11 shows that  $\beta \neq 0$  is unramified, hence  $\operatorname{Br}_{\operatorname{nr}} k(M)^G \neq 0$ .

**Theorem 8.15** (Saltman [126]). Let G be a finite group, let M be a faithful G-lattice and  $\alpha \in Z^1(G, \widehat{M})$ . Let  $k(M)_{\alpha}$  denote the field of fractions of the twisted G-k-algebra k[M]. One then has:

$$\mathrm{Br}_{\mathrm{nr}}(k(M)^G_\alpha) \cong \ker \left[ H^2(G, k[M]^*_\alpha) \xrightarrow{\mathrm{Res}} \prod_{A \in \mathcal{B}'_G} H^2(A, k^* \oplus M) \right]$$

where  $\mathcal{B}'_G$  denotes the set of bicyclic subgroups A of G such that the map

$$H^2(A, k^*) \to H^2(A, k[M]^*_\alpha)$$

is injective.

*Proof.* Once more we refer to Saltman's original paper [126] and to Barge's more natural proof [5].  $\Box$ 

**Theorem 8.16** (Barge [5]). For a finite group G, the following conditions are equivalent:

(i) For every faithful G-twisted multiplicative action  $(M, \alpha)$ , we have

$$\operatorname{Br}_{\operatorname{nr}} k(M)^G_{\alpha} = 0.$$

(ii) All Sylow subgroups of G are cyclic.

*Example* 8.17 (twisted multiplicative example). Let p be a prime and let  $G \cong \mathbb{Z}/p \times \mathbb{Z}/p$ . There exists a twisted multiplicative action of G on a lattice M such that  $\operatorname{Br}_{\operatorname{nr}} k(M)^G_{\alpha} \neq 0$ . In particular there exists a twisted multiplicative action of  $G = \mathbb{Z}/2 \times \mathbb{Z}/2$  on a lattice M such that the field of invariants  $k(M)^G_{\alpha}$  is not pure.

*Remark* 8.18. Given a *G*-lattice *M* one may show that the field  $k(M)^G$  is stably equivalent to the function field of a torus defined on the field  $k(\mathbb{Z}[G])^G$ .

For some computations on multiplicative invariants, see [7–10, 50, 51, 53, 125]. For a recent general report on multiplicative actions, we refer to the forthcoming book by M. Lorenz [86].

#### 9. Homogeneous spaces

In this section, the ground field k is the field  $\mathbb{C}$  of complex numbers. We fix an isomorphism between the group of all roots of unity in  $\mathbb{C}$  and  $\mathbb{Q}/\mathbb{Z}$ . The tools used are topological. However standard arguments show that once the statements of Theorems 9.1 and 9.13 have been proved over  $\mathbb{C}$ , they hold over any algebraically closed field  $\Omega$  of characteristic zero: reduction to an algebraically closed field k which may be embedded both in  $\mathbb{C}$  and  $\Omega$ , then invariance of étale cohomology with finite coefficients under extension of algebraically closed ground fields.

#### 9.1. The case of an almost free linear representation.

**Theorem 9.1** (Bogomolov [17, Lemma 5.7]). Let G be a connected algebraic group over the complex field  $k = \mathbb{C}$ , and let V be an almost free finite dimensional linear representation of G. Then  $\operatorname{Br}_{nr} k(V)^G = 0$ .

For  $G = PGL_n$ , this is a theorem of Saltman [121, Theorem 2.9], for which alternate proofs are given in [37, Theorem 9.7] and [127]. This answered a question of Procesi [110].

**Lemma 9.2.** For any linear algebraic group G over a field k and any positive integer s, there exists a k-linear representation V of G and a non-empty G-stable open set  $U \subset V$  such that

- (i) the complement  $V \setminus U$  is of codimension  $\geq s$ ,
- (ii) there is a morphism U → U/G of k-varieties making U into a G-torsor over U/G.

*Proof.* See Totaro [149, Remark 1.4, p. 252]. Let  $G \subset GL_n$  be any faithful k-linear representation of G. Let  $N \geq 1$  be an integer. Let  $W = \mathcal{M}_{N+n}$  be the vector space of  $(N+n) \times (N+n)$  matrices M over k:

$$M = \left( \begin{array}{c|c} A & B \\ \hline C & D \end{array} \right) \stackrel{\uparrow}{\uparrow} N$$

Let  $V = \mathcal{M}_{n,N+n}$  be the vector space of  $n \times (N+n)$  matrices over k and let  $\pi: W \to V$  be the linear projection given by

$$\left(\begin{array}{c|c} A & B \\ \hline C & D \end{array}\right) \longmapsto \left(\begin{array}{c|c} C & D \end{array}\right).$$

Let  $\Omega = \operatorname{GL}_{N+n} \subset W$  be the group of invertible matrices. We denote by  $\widetilde{G} \cong G$  the subgroup of matrices of the form

$$\left( \begin{array}{c|c} I & 0 \\ \hline 0 & D \end{array} \right)$$

where  $D \in G$  and  $I = I_N$  is the  $N \times N$  identity matrix. We denote by  $H \triangleleft \operatorname{GL}_{N+n}$  the invariant subgroup consisting of matrices of the form

$$\left(\begin{array}{c|c}
A & B \\
\hline
0 & I
\end{array}\right)$$

where  $A \in \operatorname{GL}_N$  and  $I = I_n$  is the  $n \times n$  identity matrix. The group H is clearly an extension of  $\operatorname{GL}_N$  by a unipotent invariant subgroup whose underlying kvariety is an affine space  $\mathbb{A}^{Nn}$ . The product  $\Gamma = H\widetilde{G} \cong H \rtimes G$  is a subgroup of  $\operatorname{GL}_{N+n}$ . Let  $U \subset V$  denote the open dense subset consisting of matrices of rank n. Let Z be its complement. The projection  $\pi: W \to V$  induces a surjective morphism of G-k-varieties

$$\pi_{\Omega} \colon \Omega \to U.$$

This map induces a G-isomorphism of k-varieties  $\operatorname{GL}_{N+n}/H \cong U$ . Since  $\Gamma = H \rtimes G$  is a closed subgroup of  $\operatorname{GL}_{N+n}$ , the canonical morphism

$$\operatorname{GL}_{N+n} \to \operatorname{GL}_{N+n} / \Gamma$$

factorizes through  $\pi_{\Omega}$  and induces a morphism  $\varpi$  of k-varieties which makes  $\operatorname{GL}_{N+n}/H$  into a G-torsor over  $\operatorname{GL}_{N+n}/\Gamma$ . Thus U/G exists and U is a G-torsor over U/G:

$$\begin{array}{ccc} \operatorname{GL}_{N+n}/H & \stackrel{\varpi}{\longrightarrow} & \operatorname{GL}_{N+n}/\Gamma \\ & \cong & \downarrow \\ & \cong & \downarrow \\ & U & \longrightarrow & U/G. \end{array}$$

Since  $\operatorname{codim}_V Z$  goes to infinity as N goes to infinity, we obtain  $\operatorname{codim}_V Z \ge s$  for N large.

Proof of Theorem 9.1. In view of the no-name lemma and Proposition 5.7, we may replace the linear representation V by a "better one". Namely, we take V as in Lemma 9.2 for s = 3. Thus we may now assume that we are given a linear representation V of G and a G-stable open set U with complement Z such that  $\operatorname{codim}_V Z \geq 3$  and such that there is a morphism  $U \to U/G$  which makes U into a G-torsor over U/G.

The connected group G is an extension

$$1 \to G' \to G \to T \to 1$$

of a torus T by a connected group G' without characters: X(G') = 0. Now we may first quotient U by the action of G', and then by the action of T, getting a morphism  $U/G' \to U/G$  which one checks makes U/G' into a T-torsor over U/G. Since torsors under tori are locally trivial, U/G' is birational to  $T \times U/G$ . The stability of the unramified Brauer group under pure extensions (Proposition 5.7) now implies  $\operatorname{Br}_{\mathrm{nr}}(k(V)^G) \cong \operatorname{Br}_{\mathrm{nr}}(k(V)^{G'})$ . It is thus enough to prove the theorem when G is connected and satisfies X(G) = 0, which we now assume.

Claim. For such G we now claim that

$$\operatorname{Br}_{\operatorname{nr}} k(V)^G \cong \operatorname{ker} \Big[ H^2(BG, \mathbb{Q}/\mathbb{Z}) \to \prod_{A \in \mathcal{B}_G} H^2(BA, \mathbb{Q}/\mathbb{Z}) \Big].$$
(9.1)

Here BG, resp. BA, denotes the classifying space of the topological group  $G = G(\mathbb{C})$ , resp. of the finite group A. Recall that  $\mathcal{B}_G$  denotes the set of finite bicyclic subgroups of  $G(\mathbb{C})$ .

Theorem 6.4 implies

$$\operatorname{Br}_{\operatorname{nr}}(k(V)^G) = \{ \alpha \in \operatorname{Br} k(V)^G \mid \alpha_A \in \operatorname{Br}_{\operatorname{nr}}(k(V)^A) \text{ for all } A \in \mathcal{B}_G \}.$$

Since A is abelian and the action on V is linear,  $k(V)^A$  is pure by Fischer's Theorem (Proposition 4.3), hence  $\operatorname{Br}_{\operatorname{nr}} k(V)^A = 0$  (Proposition 5.7). The field  $k(V)^G$ , resp.  $k(V)^A$ , is the function field of the smooth variety U/G, resp. U/A. Using Theorem 5.11, we get:

$$\operatorname{Br}_{\operatorname{nr}} k(V)^G \cong \operatorname{ker} \Big[ \operatorname{Br}(U/G) \to \prod_{A \in \mathcal{B}_G} \operatorname{Br}(U/A) \Big].$$
(9.2)

**Lemma 9.3.** Let X be a smooth algebraic variety over  $\mathbb{C}$ . If Pic X is torsion, then we have a canonical isomorphism

$$H^2_{\text{ét}}(X, \mathbb{Q}/\mathbb{Z}) \cong \operatorname{Br} X.$$

*Proof.* This is a consequence of "Kummer's exact sequence":

$$0 \to \operatorname{Pic} X \otimes \mathbb{Q}/\mathbb{Z} \to H^2_{\operatorname{\acute{e}t}}(X, \mathbb{Q}/\mathbb{Z}) \to \operatorname{Br} X \to 0.$$

**Lemma 9.4.** Let X be a smooth connected variety over  $k = \mathbb{C}$ . Let G be a linear algebraic group with character group  $X(G) = \text{Hom}(G, \mathbb{G}_m)$ . Let  $X \to X/G$  be a G-torsor.

(i) If G is connected, there is a natural exact sequence

$$1 \to k[X/G]^* \to k[X]^* \to X(G) \to \operatorname{Pic}(X/G) \to \operatorname{Pic} X.$$

(ii) If G is a finite constant group A, there is a natural exact sequence

$$1 \to X(A) \to \operatorname{Pic}(X/A) \to \operatorname{Pic} X$$

*Proof.* For (i), see [134, Proposition 6.10]. For (ii), the same arguments yield an exact sequence

$$1 \to \operatorname{Hom}(A, k[X]^*) \to \operatorname{Pic}(X/A) \to \operatorname{Pic} X.$$

Then use the fact that  $k[X]^*/k^*$  is torsionfree.

**Lemma 9.5.** Let X be a smooth connected variety over  $\mathbb{C}$ . Let G be a linear algebraic group. Let  $X \to X/G$  be a G-torsor. Assume that Pic X is a torsion group and that G is either finite, or connected and characterfree. Then we have a canonical isomorphism

$$\operatorname{Br}(X/G) \cong H^2_{\operatorname{\acute{e}t}}(X/G, \mathbb{Q}/\mathbb{Z}).$$

*Proof.* Under our assumptions, Lemma 9.4 implies that  $\operatorname{Pic}(X/G)$  is torsion. Then Lemma 9.3 gives the isomorphism  $\operatorname{Br}(U/G) \cong H^2(U/G, \mathbb{Q}/\mathbb{Z})$ .  $\Box$ 

*Proof of Theorem 9.1 (continued).* Lemma 9.5 applied to X = U together with (9.2) implies

$$\operatorname{Br}_{\operatorname{nr}} k(V)^G \cong \operatorname{ker} \Big[ H^2(U/G, \mathbb{Q}/\mathbb{Z}) \to \prod_{A \in \mathcal{B}_G} H^2(U/A, \mathbb{Q}/\mathbb{Z}) \Big].$$

Since  $U \to U/G$  is a *G*-torsor, there is a map of topological spaces  $U/G \to BG$ such that  $U \to U/G$  is the pull-back of the universal covering space  $EG \to BG$ . Similarly, there is a map  $U/A \to BA$ . Now the assumption on the codimension of the complement of U in V, which had not been used yet, implies that the maps  $U/G \to BG$  and  $U/A \to BA$  induce isomorphisms on the cohomology

groups  $H^i_{\text{\acute{e}t}}(\ \cdot\ ,\mathbb{Q}/\mathbb{Z})$  for  $i \leq 2$  (the projections  $U \to U/G$  and  $U \to U/A$  are algebraic approximations of the universal covering spaces). The "claim" (9.1) now follows.

**Lemma 9.6.** If  $\widetilde{G}$  is a connected, simply connected group, then

$$H^1(B\widetilde{G}, \mathbb{Q}/\mathbb{Z}) = H^2(B\widetilde{G}, \mathbb{Q}/\mathbb{Z}) = 0.$$

Proof. The fibration  $E\tilde{G} \to B\tilde{G}$  yields isomorphisms  $\pi_1 B\tilde{G} = \pi_0 \tilde{G} = 0$  (since  $\tilde{G}$  is connected) and  $\pi_2 B\tilde{G} = \pi_1 \tilde{G} = 0$  (since  $\tilde{G}$  is simply connected). Thus  $B\tilde{G}$  is 2-connected. The Hurewicz theorem now yields  $H_1(B\tilde{G}) = H_2(B\tilde{G}) = 0$ . The universal coefficient theorem now yields the statement of the lemma.  $\Box$ 

**Lemma 9.7.** If G is a connected group without characters, the universal covering  $\widetilde{G} \to G$  defines a natural isomorphism of finite abelian groups

$$d_{2,G} \colon \operatorname{Hom}(\pi_1 G, \mathbb{Q}/\mathbb{Z}) \xrightarrow{\cong} H^2(BG, \mathbb{Q}/\mathbb{Z}).$$

*Proof.* Since G is a connected group without characters,  $G = R_{\rm u}(G) \rtimes G_{\rm ss}$ with  $R_{\rm u}(G)$  unipotent and  $G_{\rm ss}$  semisimple. Thus its fundamental group  $\pi_1 G$ is a finite abelian group. We have the Leray-Serre spectral sequence [91, §5, Theorem 5.2] for the fibration  $\pi_1 G \to B\widetilde{G} \to BG$ :

$$E_2^{p,q} = H^p(BG, \mathcal{H}^q(\pi_1 G, \mathbb{Q}/\mathbb{Z})) \implies H^{p+q}(B\widetilde{G}, \mathbb{Q}/\mathbb{Z}).$$

Since BG is 1-connected, each local system  $\mathcal{H}^q(\pi_1 G, \mathbb{Q}/\mathbb{Z})$  is constant. Then the associated exact sequence of terms of lower degree

$$H^1(B\widetilde{G}, \mathbb{Q}/\mathbb{Z}) \to H^1(\pi_1 G, \mathbb{Q}/\mathbb{Z}) \xrightarrow{d_{2,G}} H^2(BG, \mathbb{Q}/\mathbb{Z}) \to H^2(B\widetilde{G}, \mathbb{Q}/\mathbb{Z})$$

together with Lemma 9.6 yields a natural isomorphism of finite abelian groups  $d_{2,G}$ : Hom $(\pi_1 G, \mathbb{Q}/\mathbb{Z}) \xrightarrow{\cong} H^2(BG, \mathbb{Q}/\mathbb{Z})$ .

**Lemma 9.8.** Let A be a finite group and C be a trivial A-module. Let

$$1 \to \pi \to \widetilde{A} \to A \to 1 \tag{$\mathcal{E}_A$}$$

be a central extension. The associated Lyndon-Hochschild-Serre spectral sequence defines a natural morphism

$$d_{2,\mathcal{E}_A} \colon \operatorname{Hom}(\pi, C) \to H^2(A, C)$$

sending  $\phi: \pi \to C$  to the class of the extension  $\phi_*(\mathcal{E}_A)$  obtained from  $(\mathcal{E}_A)$  by push-out along  $\phi$ .

*Proof.* The Leray-Serre spectral sequence [91, §5, Theorem 5.2] for the fibration  $\pi \to B\widetilde{A} \to BA$  associated to  $(\mathcal{E}_A)$ :

$$E_2^{p,q} = H^p(BA, \mathcal{H}^q(\pi, C)) \implies H^{p+q}(B\widetilde{A}, C)$$

coincides with the Lyndon-Hochschild-Serre spectral sequence (see [91,  $\S 8^{bis}.2$ , p. 342])

$$E_2^{p,q} = H^p(A, H^q(\pi, C)) \implies H^{p+q}(\widetilde{A}, C)$$

associated to  $(\mathcal{E}_A)$ . This extension being central, the action of A on  $H^q(\pi, C)$  is trivial and the local system  $\mathcal{H}^q(\pi, C)$  is constant on BA. Hence both

morphisms  $H^0(BA, \mathcal{H}^1(\pi, C)) \to H^2(BA, \mathcal{H}^0(\pi, C))$  and  $H^0(A, H^1(\pi, C)) \to H^2(A, H^0(\pi, C))$  coincide, yielding the natural morphism

$$d_{2,\mathcal{E}_A} \colon \operatorname{Hom}(\pi,C) \to H^2(A,C).$$

It is well known (see [26, §IV.3, Theorem 3.12]) that  $H^2(A, C)$  classifies abstract central extensions of A by C. Let  $C = \pi$ . One can check that  $d_{2,\mathcal{E}_A}(\mathrm{id}_{\pi})$ is the class of the central extension  $\mathcal{E}_A$ . Now for any C, the functoriality of the LHS spectral sequence implies that  $d_{2,\mathcal{E}_A}(\phi)$  is the class of  $\phi_*(\mathcal{E}_A)$ .  $\Box$ 

Proof of Theorem 9.1 (continued). Let  $R_{u}(G)$  be the unipotent radical of G. We have the exact sequence

$$1 \to R_{\rm u}(G) \to G \xrightarrow{\varrho} G_1 \to 1, \tag{9.3}$$

where  $G_1$  is semisimple. Pulling back the universal cover  $\widetilde{G}_1 \to G_1$  over G, we get the commutative diagram of exact sequences

where  $\widetilde{G} \to G$  is the universal cover of G. From Lemma 9.7 we get the commutative diagram

$$\operatorname{Hom}(\pi_1 G, \mathbb{Q}/\mathbb{Z}) \xrightarrow{\cong} H^2(BG, \mathbb{Q}/\mathbb{Z})$$
$$\uparrow \qquad \qquad \uparrow$$
$$\operatorname{Hom}(\pi_1 G_1, \mathbb{Q}/\mathbb{Z}) \xrightarrow{\cong} H^2(BG_1, \mathbb{Q}/\mathbb{Z}).$$

Since  $R_{u}(G)$  is contractible, the map  $\pi_{1}G \to \pi_{1}G_{1}$  induced by (9.3) is an isomorphism. Hence the map  $H^{2}(BG_{1}, \mathbb{Q}/\mathbb{Z}) \to H^{2}(BG, \mathbb{Q}/\mathbb{Z})$  is an isomorphism. For each finite subgroup  $A \subset G$ , the exact sequence (9.3) induces an isomorphism  $A \cong \varrho(A)$ . Moreover it induces a bijection  $A \mapsto \varrho(A)$  between the finite subgroups of G and those of  $G_{1}$ . We conclude that

$$\ker \Big[ H^2(BG, \mathbb{Q}/\mathbb{Z}) \to \prod_{A \in \mathcal{B}_G} H^2(BA, \mathbb{Q}/\mathbb{Z}) \Big]$$

and

$$\ker \Big[ H^2(BG_1, \mathbb{Q}/\mathbb{Z}) \to \prod_{A \in \mathcal{B}_{G_1}} H^2(BA, \mathbb{Q}/\mathbb{Z}) \Big]$$

are isomorphic. To prove that the first kernel is trivial we may therefore assume that G is a semisimple group.

Consider now the natural isogeny  $\widetilde{G} \to G$  where  $\widetilde{G}$  denotes the simply connected cover of G. Each  $A \in \mathcal{B}_G$  gives rise to a commutative diagram of exact sequences of groups

Then (9.4) gives a morphism of Leray-Serre spectral sequences:

hence  $d_2 \colon E_2^{0,1} \to E_2^{2,0}$  defines a commutative square (see Lemmas 9.7 and 9.8)

Any element  $\alpha \in H^2(BG, \mathbb{Q}/\mathbb{Z})$  may be interpreted as a homomorphism  $\phi$  of  $\pi_1 G = \pi_1^{\text{alg}} G$  to  $\mathbb{Q}/\mathbb{Z}$ , hence to some  $\mathbb{Z}/n = \text{Im }\phi$ . Hence by push-out from the natural extension

$$1 \to \pi_1 G \to \widetilde{G} \to G \to 1$$

it defines an extension

$$1 \to \mathbb{Z}/n \to G_1 \to G \to 1,$$

where  $G_1$  is a connected semisimple algebraic group over  $\mathbb{C}$ . This extension itself defines an extension of G by  $\mathbb{Q}/\mathbb{Z}$ . By Lemma 9.8, the natural restriction map  $r: H^2(BG, \mathbb{Q}/\mathbb{Z}) \to H^2(BA, \mathbb{Q}/\mathbb{Z})$  may be interpreted in terms of restrictions of extensions. More precisely, if  $\alpha \in H^2(BG, \mathbb{Q}/\mathbb{Z})$  corresponds to the isogeny

$$1 \to \mathbb{Z}/n \to G_1 \to G \to 1$$

(notations as above), and if  $A \subset G$  is a finite subgroup, then the restriction of the above isogeny to this subgroup gives rise to a central extension

$$1 \to \mathbb{Z}/n \to A_1 \to A \to 1$$

whose class in  $H^2(A, \mathbb{Z}/n)$  restricts to the class of  $r(\alpha) \in H^2(A, \mathbb{Q}/\mathbb{Z}) \cong H^2(BA, \mathbb{Q}/\mathbb{Z})$ .

To prove that

$$\ker \Big[ H^2(BG, \mathbb{Q}/\mathbb{Z}) \to \prod_{A \in \mathcal{B}_G} H^2(BA, \mathbb{Q}/\mathbb{Z}) \Big] = 0,$$

which will complete the proof of the theorem, all we now need to prove is that if we are given a nontrivial central extension

$$1 \to \mathbb{Z}/n \to G_1 \to G \to 1 \tag{9.6}$$

with  $G_1$  connected, then there exists a finite bicyclic subgroup A in the semisimple group G such that the restriction

$$1 \to \mathbb{Z}/n \to A_1 \to A \to 1 \tag{9.7}$$

of (9.6) to A is a nontrivial extension. Since (9.7) is a central extension, it is nontrivial if  $A_1$  is not commutative.

Let us postpone the proof of the:

**Proposition 9.9.** Any element in the centre of a connected semisimple group is a commutator of elements of finite order.

Proof of Theorem 9.1 (end). Let c be a generator of the subgroup  $\mathbb{Z}/n$  of the centre of the connected semisimple group  $G_1$ . Using Proposition 9.9 we can write c as a commutator  $a_1b_1a_1^{-1}b_1^{-1}$ , with  $a_1$  and  $b_1$  of finite order in  $G_1$ , let a and b be the images of  $a_1$  and  $b_1$  in G, and let  $A \subset G$  be the finite, abelian, bicyclic group which they generate. Since  $a_1$  and  $b_1$  do not commute, the group  $A_1$  is not commutative and the proof of Theorem 9.1 is complete.  $\Box$ 

Proof of Proposition 9.9. It is enough to give the proof when the connected semisimple group G is simply connected, and then it is enough to prove it when G is simple and simply connected.

First proof (suggested by O. Gabber). To each choice of a basis of roots in the character group X(T) of a maximal torus  $T \subset G$  one may associate an element c of the Weyl group W of G, known as the Coxeter element [25, V.6.2 and VI.1.11].

It is known (loc. cit.) that 1 is not an eigenvalue of c for its action on  $\mathfrak{t} = X(T) \otimes \mathbb{C}$ . Let d in the normalizer N(T) of T be a representant of c. Such a representant may be chosen of finite order. Conjugacy by d on T induces an automorphism whose tangent linear map is  $c \in \operatorname{Aut} \mathfrak{t}$ . Now the tangent map to the homomorphism  $\lambda: T \to T$  given by  $\lambda(t) = dtd^{-1}t^{-1}$  is  $c - \operatorname{id} \in \operatorname{End} \mathfrak{t}$ , hence is invertible. Thus  $\lambda$  is an isogeny, and this isogeny induces a surjection on torsion points  $\lambda: T_{\operatorname{tors}} \to T_{\operatorname{tors}}$ . We conclude that any element  $x \in T_{\operatorname{tors}}$  may be written as  $x = dyd^{-1}y^{-1}$  in G, with  $y \in T_{\operatorname{tors}}$ . Since the centre of G is contained in any maximal torus of G, the conclusion follows.

Second proof (suggested by J-P. Serre). One first proves the result for  $G = \operatorname{SL}_n$  (a slick proof being as follows: the algebra generated by a and b with the relations  $a^n = b^n = 1$  and  $ab = \zeta ba$  with  $\zeta^n = 1$  is none other than  $\mathcal{M}_n(\mathbb{C})$ ). Now inspection of the root systems reveals that any simply connected semisimple group G contains a subgroup  $H \cong \operatorname{SL}_{n_1} \times \cdots \times \operatorname{SL}_{n_r}$  with rank  $H = \operatorname{rank} G$ , hence with  $\operatorname{centre}(G) \subset \operatorname{centre}(H)$ .

9.2. The case of a homogeneous space. For Theorem 9.13, we need some preparation. Let us first recall a theorem of Steinberg (cf. [144, II, 3.9, p. 197]).

**Theorem 9.10** (Steinberg). If G is a semisimple simply connected group, the centralizer of a semisimple element of G is a connected reductive group.

*Remark.* This is no longer true if G is not simply connected.

**Corollary 9.11.** Given two commuting semisimple elements in a semisimple, simply connected group G, there is a maximal torus of G which contains them both.

*Proof.* Let x, y be two such elements. Let  $H = Z_G(x)$  be the centralizer of x. Then H is a connected reductive group (Theorem 9.10), and x is in its center. Hence x belongs to any maximal torus of H. The element y belongs to H. Since it is semisimple, it belongs to a maximal torus T of H. There exists a maximal torus of G which contains T. Such a torus contains both x and y.

**Proposition 9.12.** Let A be a bicyclic finite subgroup of a simply connected group G. Then G/A is a rational variety, and the unramified Brauer group of G/A is trivial.

*Proof.* We have the natural exact sequence

$$1 \to U \to G \xrightarrow{\pi} G_{\rm ss} \to 1,$$

where U is the unipotent radical of G and  $G_{ss}$  is a connected, semisimple, simply connected group. Since  $U \cap A = \{1\}$ , the map  $A \to \pi(A)$  is an isomorphism and the natural projection  $G/A \to G_{ss}/A$  defines a U-torsor. Hence the varieties G/A and  $G_{ss}/A \times U$  are isomorphic. To prove the proposition we may therefore assume that G is semisimple and simply connected.

Let  $A \subset G$  be a finite bicyclic group. By the previous corollary, there is a maximal torus T which contains A. Let T' = T/A. The map  $G \to G/T$ makes G into a T-torsor over G/T, hence G is birational to  $T \times G/T$ . On the other hand, the map  $G/A \to G/T$  makes G/A into a T'-torsor over G/T, hence G/A is birational to  $T' \times G/T$ . Since T and T' are birational to each other, we conclude that G/A is birational to G, hence is a rational variety.  $\Box$ 

The proof of the following Theorem uses Theorem 9.1. In turn, Proposition 4.9 shows that Theorem 9.13 generalizes Theorem 9.1.

**Theorem 9.13** (Bogomolov [18, Theorem 2.4]). Let G be a connected, simply connected group over  $\mathbb{C}$ , and let  $H \subset G$  be a connected closed subgroup. Then the unramified Brauer group of G/H vanishes.

*Proof.* There is a closed normal subgroup  $H_1 \subset H$  which is connected and characterfree, such that  $H/H_1$  is a torus T. Thus the map  $G/H_1 \to G/H$  makes the first variety into a T-torsor over the second one. Since torsors under tori are locally trivial,  $G/H_1$  is birational to  $T \times G/H$ . By Proposition 5.7 it is thus enough to prove the theorem when H is connected and characterfree, which we now assume.

Just as in the proof of Theorem 9.1, we have an exact sequence:

$$0 \to \operatorname{Pic}(G/H) \otimes \mathbb{Q}/\mathbb{Z} \to H^2_{\operatorname{\acute{e}t}}(G/H, \mathbb{Q}/\mathbb{Z}) \to \operatorname{Br}(G/H) \to 0$$

and similar exact sequences for any finite subgroup  $A \subset H$ :

$$0 \to \operatorname{Pic}(G/A) \otimes \mathbb{Q}/\mathbb{Z} \to H^2_{\operatorname{\acute{e}t}}(G/A, \mathbb{Q}/\mathbb{Z}) \to \operatorname{Br}(G/A) \to 0$$

We also have exact sequences (Lemma 9.4)

 $X(H) \to \operatorname{Pic}(G/H) \to \operatorname{Pic} G$ 

and

$$X(A) \to \operatorname{Pic}(G/A) \to \operatorname{Pic} G.$$

Now Pic G = 0 because G is connected and simply connected, X(H) = 0, and X(A) is finite since A is finite. We thus get a commutative diagram:

Proposition 9.12 together with the general formula for the unramified Brauer group (Theorem 6.4) and the above diagram imply that

$$\operatorname{Br}_{\operatorname{nr}} G/H \cong \operatorname{ker} \Big[ H^2(G/H, \mathbb{Q}/\mathbb{Z}) \to \prod_{A \in \mathcal{B}_H} H^2(G/A, \mathbb{Q}/\mathbb{Z}) \Big].$$
(9.9)

From the universal property of the topological *H*-fibration  $EH \rightarrow BH$  there is a cartesian diagram of topological morphisms

$$\begin{array}{ccc} G & \longrightarrow & EH \\ & & & \downarrow \\ G/H & \longrightarrow & BH. \end{array}$$

This induces a map of spectral sequences

hence a commutative diagram:

$$\begin{array}{ccc} H^{0}(G/H,\mathcal{H}^{1}(H,\mathbb{Q}/\mathbb{Z})) & \stackrel{\partial}{\longrightarrow} & H^{2}(G/H,\mathcal{H}^{0}(H,\mathbb{Q}/\mathbb{Z})) \\ & \uparrow & & \uparrow \\ & & & \uparrow \\ H^{0}(BH,\mathcal{H}^{1}(H,\mathbb{Q}/\mathbb{Z})) & \stackrel{\partial}{\longrightarrow} & H^{2}(BH,\mathcal{H}^{0}(H,\mathbb{Q}/\mathbb{Z})). \end{array}$$

Since G is simply connected,  $\pi_1 G = 0$ . It is also known (E. Cartan's theorem) that  $\pi_2 G = 0$ . Thus  $H^1(G, \mathbb{Q}/\mathbb{Z}) = 0$  and the universal coefficient theorem also gives  $H^2(G, \mathbb{Q}/\mathbb{Z}) = 0$ . We clearly have the same vanishing properties for EH, which is contractible. This implies that the maps  $\partial$  are isomorphisms.

Both G/H and BH are connected and simply connected, as follows from the long sequence of homotopy groups deduced from the two fibrations (H is connected), hence each local system  $\mathcal{H}^q(H, \mathbb{Q}/\mathbb{Z})$  is constant and equal to  $H^q(H, \mathbb{Q}/\mathbb{Z})$ . Since BH and G/H are connected, the L.H.S. vertical map is identity on  $H^1(H, \mathbb{Q}/\mathbb{Z})$ , hence the R.H.S. vertical map is an isomorphism. But this map is the natural map  $H^2(BH, \mathbb{Q}/\mathbb{Z}) \to H^2(G/H, \mathbb{Q}/\mathbb{Z})$ .

*Remark.* A shorter proof runs as follows. We may regard the map  $G/H \to BH$  as a topological fibration with fibre G. Indeed the map  $G/H \times EG \to (EG)/H$  makes  $G/H \times EG$  into a principal G-bundle over (EG)/H. The total space  $G/H \times EG$  has the same homotopy type as G/H and the base (EG)/H has the same homotopy type as BH. Associated to this fibration there is a spectral sequence

$$E_2^{p,q} = H^p(BH, H^q(G, \mathbb{Q}/\mathbb{Z})) \implies H^{p+q}(G/H, \mathbb{Q}/\mathbb{Z}).$$

and this spectral sequence yields an isomorphism

$$H^2(BH, \mathbb{Q}/\mathbb{Z}) \xrightarrow{\cong} H^2(G/H, \mathbb{Q}/\mathbb{Z})$$

which is induced by the map  $G/H \to BH$ .

From the universal property of the topological A-covering  $EA \rightarrow BA$  there is a cartesian diagram of topological morphisms

$$\begin{array}{ccc} G & \longrightarrow & EA \\ & & & \downarrow \\ G/A & \longrightarrow & BA. \end{array}$$

This induces a map of spectral sequences

Since  $H^1(G, \mathbb{Q}/\mathbb{Z})$  and  $H^2(G, \mathbb{Q}/\mathbb{Z})$  are zero, and also all  $H^q(EA, \mathbb{Q}/\mathbb{Z})$  for q > 0, we get the commutative diagram:

$$\begin{array}{cccc} H^{2}(A, H^{0}(G, \mathbb{Q}/\mathbb{Z})) & \stackrel{\cong}{\longrightarrow} & H^{2}(G/A, \mathbb{Q}/\mathbb{Z})) \\ & \uparrow & & \uparrow \\ H^{2}(A, H^{0}(EA, \mathbb{Q}/\mathbb{Z})) & \stackrel{\cong}{\longrightarrow} & H^{2}(BA, \mathbb{Q}/\mathbb{Z})). \end{array}$$

Since G and EA are connected, the L.H.S. vertical map is identity on the group  $H^2(A, \mathbb{Q}/\mathbb{Z})$ , hence the R.H.S. vertical map is an isomorphism:

$$H^2(BA, \mathbb{Q}/\mathbb{Z}) \xrightarrow{\cong} H^2(G/A, \mathbb{Q}/\mathbb{Z}).$$

Using the diagram

$$\begin{array}{cccc} G/A & \longrightarrow & BA \\ & & & \downarrow \\ G/H & \longrightarrow & BH \end{array}$$

we get the commutative diagram

$$\begin{array}{cccc} H^2(G/H, \mathbb{Q}/\mathbb{Z}) & \longrightarrow & H^2(G/A, \mathbb{Q}/\mathbb{Z})) \\ & & & \uparrow & & \\ H^2(BH, \mathbb{Q}/\mathbb{Z}) & \longrightarrow & H^2(BA, \mathbb{Q}/\mathbb{Z})). \end{array}$$

We have proved that both vertical maps are isomorphisms. Formula (9.9) now yields:

$$\operatorname{Br}_{\operatorname{nr}} G/H \cong \operatorname{ker} \Big[ H^2(BH, \mathbb{Q}/\mathbb{Z}) \to \prod_{A \in \mathcal{B}_H} H^2(BA, \mathbb{Q}/\mathbb{Z}) \Big],$$

and we saw in the proof of Theorem 9.1 that this kernel is zero.

*Remark* 9.14. It would be interesting to know whether Theorem 9.13 extends to quotients G/H with G and H connected.

Note that if G is connected, semisimple but not simply connected, and if  $\widetilde{G}$  is its simply connected cover, the inverse image of H in  $\widetilde{G}$  need not be connected, so that one may not reduce to the situation of the theorem. For example, if we consider in  $\widetilde{G} = \mathrm{SL}(3)$  the subgroup  $H_1$  which is the product of  $\{1\} \times \mathrm{SL}(2)$  by the diagonal  $\mu_3$ , the image H of that group in the projective special linear group PSL(3) is clearly connected whereas  $H_1$  is not connected.

#### References

- H. Ahmad, M. Hajja, and M.-C. Kang, *Rationality of some projective linear actions*, J. Algebra **228** (2000), no. 2, 643–658.
- M. Artin, On Azumaya algebras and finite dimensional representations of rings., J. Algebra 11 (1969), 532–563.
- [3] M. Artin and D. Mumford, Some elementary examples of unirational varieties which are not rational, Proc. London Math. Soc. (3) 25 (1972), 75–95.
- [4] J. Barge, Cohomologie des groupes et corps d'invariants multiplicatifs, Math. Ann. 283 (1989), no. 3, 519–528.
- [5] \_\_\_\_\_, Cohomologie des groupes et corps d'invariants multiplicatifs tordus, Comment. Math. Helv. 72 (1997), no. 1, 1–15.
- [6] A. Beauville, J.-L. Colliot-Thélène, J.-J. Sansuc, and P. Swinnerton-Dyer, Variétés stablement rationnelles non rationnelles, Ann. of Math. (2) **121** (1985), no. 2, 283– 318.
- [7] E. Beneish, Induction theorems on the stable rationality of the center of the ring of generic matrices, Trans. Amer. Math. Soc. 350 (1998), no. 9, 3571–3585.
- [8] \_\_\_\_\_, Noether settings for central extensions of groups with zero Schur multiplier, J. Algebra Appl. 1 (2002), no. 1, 107–112.
- [9] \_\_\_\_\_, The center of the generic division ring and twisted multiplicative group actions, J. Algebra 259 (2003), no. 2, 313–322.
- [10] \_\_\_\_\_, Monomial actions of the symmetric group, J. Algebra 265 (2003), no. 2, 405– 419.
- [11] A. Berele and D. J. Saltman, The centers of generic division algebras with involution, Israel J. Math. 63 (1988), no. 1, 98–118.
- [12] C. Bessenrodt and L. Le Bruyn, Stable rationality of certain  $PGL_n$ -quotients, Invent. Math. **104** (1991), no. 1, 179–199.
- [13] A. Białynicki-Birula, Categorical quotients, J. Algebra 239 (2001), no. 1, 35–55.
- [14] \_\_\_\_\_, Quotients by actions of groups, Algebraic quotients. Torus actions and cohomology. The adjoint representation and the adjoint action, 2002, pp. 1–82.

#### J.-L. COLLIOT-THÉLÈNE AND J.-J. SANSUC

- [15] F. A. Bogomolov, Rationality of the moduli of hyperelliptic curves of arbitrary genus, Proceedings of the 1984 Vancouver conference in algebraic geometry, 1986, pp. 17–37.
- [16] \_\_\_\_\_, Stable rationality of quotient spaces for simply connected groups, Mat. Sb. (N.S.) 130(172) (1986), no. 1, 3–17, 128; English transl., Math. USSR-Sb. 58 (1987), no. 1, 1–14.
- [17] \_\_\_\_\_, The Brauer group of quotient spaces of linear representations, Izv. Akad. Nauk SSSR Ser. Mat. 51 (1987), no. 3, 485–516, 688; English transl., Math. USSR-Izv. 30 (1988), no. 3, 455–485.
- [18] \_\_\_\_\_, Brauer groups of the fields of invariants of algebraic groups, Mat. Sb. 180 (1989), no. 2, 279–293; English transl., Math. USSR-Sb. 66 (1990), no. 1, 285–299.
- [19] \_\_\_\_\_, Stable cohomology of groups and algebraic varieties, Mat. Sb. 183 (1992), no. 5, 3–28; English transl., Russian Acad. Sci. Sb. Math. 76 (1993), no. 1, 1–21.
- [20] F. A. Bogomolov and P. I. Katsylo, *Rationality of some quotient varieties*, Mat. Sb. (N.S.) **126(168)** (1985), no. 4, 584–589; English transl., Math. USSR-Sb. **54** (1986), no. 4, 571–576.
- [21] F. Bogomolov, J. Maciel, and T. Petrov, Unramified Brauer groups of finite simple groups of Lie type A<sub>l</sub>, Amer. J. Math. **126** (2004), no. 4, 935–949.
- [22] F. Bogomolov, T. Petrov, and Y. Tschinkel, Rationality of moduli of elliptic fibrations with fixed monodromy, Geom. Funct. Anal. 12 (2002), no. 6, 1105–1160.
- [23] A. Borel, *Linear algebraic groups*, 2nd ed., Graduate Texts in Mathematics, vol. 126, Springer-Verlag, New York, 1991.
- [24] M. Borovoi and B. Kunyavskii, Formulas for the unramified Brauer group of a principal homogeneous space of a linear algebraic group, J. Algebra 225 (2000), no. 2, 804–821.
- [25] N. Bourbaki, Éléments de mathématique. Fasc. XXXIV. Groupes et algèbres de Lie. Chapitre IV: Groupes de Coxeter et systèmes de Tits. Chapitre V: Groupes engendrés par des réflexions. Chapitre VI: systèmes de racines, Actualités Scientifiques et Industrielles, No. 1337, Hermann, Paris, 1968.
- [26] K. S. Brown, Cohomology of groups, Graduate Texts in Mathematics, vol. 87, Springer-Verlag, New York, 1994. Corrected reprint of the 1982 original.
- [27] V. Chernousov, Ph. Gille, and Z. Reichstein, Resolving G-torsors by abelian base extensions, J. Algebra, to appear, available at http://www.math.ubc.ca/~reichst/ torsor-ind.html.
- [28] Séminaire C. Chevalley; 2e année: 1958. Anneaux de Chow et applications, Secrétariat mathématique, 11 rue Pierre Curie, Paris, 1958.
- [29] H. Chu, S.-J. Hu, and M.-C. Kang, Noether's problem for dihedral 2-groups, Comment. Math. Helv. 79 (2004), no. 1, 147–159.
- [30] H. Chu and M.-C. Kang, *Rationality of p-group actions*, J. Algebra 237 (2001), no. 2, 673–690.
- [31] C. H. Clemens and P. A. Griffiths, The intermediate Jacobian of the cubic threefold, Ann. of Math. (2) 95 (1972), 281–356.
- [32] J.-L. Colliot-Thélène, Birational invariants, purity and the Gersten conjecture, Ktheory and algebraic geometry: connections with quadratic forms and division algebras (Santa Barbara, CA, 1992), 1995, pp. 1–64.
- [33] J.-L. Colliot-Thélène and B. È. Kunyavskiĭ, Groupe de Brauer non ramifié des espaces principaux homogènes de groupes linéaires, J. Ramanujan Math. Soc. 13 (1998), no. 1, 37–49.
- [34] \_\_\_\_\_, Groupe de Picard et groupe de Brauer des compactifications lisses d'espaces homogènes, in preparation.
- [35] J.-L. Colliot-Thélène and M. Ojanguren, Variétés unirationnelles non rationnelles: audelà de l'exemple d'Artin et Mumford, Invent. Math. 97 (1989), no. 1, 141–158.
- [36] J.-L. Colliot-Thélène and J.-J. Sansuc, La R-équivalence sur les tores, Ann. Sci. École Norm. Sup. (4) 10 (1977), no. 2, 175–229.
- [37] \_\_\_\_\_, Principal homogeneous spaces under flasque tori: applications, J. Algebra 106 (1987), no. 1, 148–205.

- [38] T. de Fernex, L. Ein, and M. Mustaţă, Bounds for log canonical thresholds with applications to birational rigidity, Math. Res. Lett. 10 (2003), no. 2-3, 219–236.
- [39] P. Deligne, Cohomologie étale, Springer-Verlag, Berlin, 1977. Séminaire de Géométrie Algébrique du Bois-Marie SGA 4<sup>1</sup>/<sub>2</sub>; avec la collaboration de J.-F. Boutot, A. Grothendieck, L. Illusie et J.-L. Verdier Lecture Notes in Mathematics, Vol. 569.
- [40] J. Dixmier and M. Raynaud, Sur le quotient d'une variété algébrique par un groupe algébrique, Mathematical analysis and applications, Part A, 1981, pp. 327–344.
- [41] I. V. Dolgachev, Rationality of fields of invariants, Algebraic geometry, Bowdoin, 1985 (Brunswick, Maine, 1985), 1987, pp. 3–16.
- [42] E. Fischer, Die Isomorphie der Invariantenkörper der endlichen Abelschen Gruppen linearen Transformationen, Nachr. König. Ges. Wiss. Göttingen (1915), 77–80.
- [43] E. Formanek, The center of the ring of 3 × 3 generic matrices, Linear and Multilinear Algebra 7 (1979), no. 3, 203–212.
- [44] \_\_\_\_\_, The center of the ring of  $4 \times 4$  generic matrices, J. Algebra **62** (1980), no. 2, 304–319.
- [45] \_\_\_\_\_, Rational function fields. Noether's problem and related questions, J. Pure Appl. Algebra **31** (1984), no. 1-3, 28–36.
- [46] V. Gatti and E. Viniberghi, Spinors of 13-dimensional space, Adv. in Math. 30 (1978), no. 2, 137–155.
- [47] A. Grothendieck, Le groupe de Brauer. I. Algèbres d'Azumaya et interprétations diverses, Dix Exposés sur la Cohomologie des Schémas, 1968, pp. 46–66.
- [48] \_\_\_\_\_, Le groupe de Brauer. II. Théorie cohomologique, Dix Exposés sur la Cohomologie des Schémas, 1968, pp. 67–87.
- [49] \_\_\_\_\_, Le groupe de Brauer. III. Exemples et compléments, Dix Exposés sur la Cohomologie des Schémas, 1968, pp. 88–188.
- [50] M. Hajja and M.-C. Kang, Finite group actions on rational function fields, J. Algebra 149 (1992), no. 1, 139–154.
- [51] \_\_\_\_\_, Three-dimensional purely monomial group actions, J. Algebra 170 (1994), no. 3, 805–860.
- [52] M. Hajja and M. C. Kang, Some actions of symmetric groups, J. Algebra 177 (1995), no. 2, 511–535.
- [53] M. Hajja and M.-C. Kang, Twisted actions of symmetric groups, J. Algebra 188 (1997), no. 2, 626–647.
- [54] A. V. Iltyakov, On rational invariants of the group  $E_6$ , Proc. Amer. Math. Soc. **124** (1996), no. 12, 3637–3640.
- [55] A. V. Iltyakov and I. P. Shestakov, On invariants of  $F_4$  and the center of the Albert algebra, J. Algebra **179** (1996), no. 3, 838–851.
- [56] V. A. Iskovskikh, Birational rigidity of Fano hypersurfaces in the framework of Mori theory, Uspekhi Mat. Nauk 56 (2001), no. 2(338), 3–86; English transl., Russian Math. Surveys 56 (2001), no. 2, 207–291.
- [57] V. A. Iskovskikh and Ju. I. Manin, Three-dimensional quartics and counterexamples to the Lüroth problem, Mat. Sb. (N.S.) 86(128) (1971), 140–166; English transl., Math. USSR-Sb. 15 (1971), 141–166.
- [58] V. G. Kac, V. L. Popov, and È. B. Vinberg, Sur les groupes linéaires algébriques dont l'algèbre des invariants est libre, C. R. Acad. Sci. Paris Sér. A-B 283 (1976), no. 12, Ai, A875–A878.
- [59] B. Kahn, Formes quadratiques sur les corps, book in preparation.
- [60] M.-C. Kang, Rationality problem of GL<sub>4</sub> group actions, Adv. Math. 181 (2004), no. 2, 321–352.
- [61] P. I. Katsylo, Rationality of orbit spaces of irreducible representations of the group SL<sub>2</sub>, Izv. Akad. Nauk SSSR Ser. Mat. **47** (1983), no. 1, 26–36; English transl., Math. USSR-Izv. **22** (1984), no. 1, 23–32.

- [62] \_\_\_\_\_, Rationality of the moduli spaces of hyperelliptic curves, Izv. Akad. Nauk SSSR Ser. Mat. 48 (1984), no. 4, 705–710; English transl., Math. USSR-Izv. 25 (1985), no. 1, 45–50.
- [63] \_\_\_\_\_, Rationality of fields of invariants of reducible representations of the group SL<sub>2</sub>, Vestnik Moskov. Univ. Ser. I Mat. Mekh. (1984), no. 5, 77–79; English transl., Moscow Univ. Math. Bull. **39** (1984), no. 5, 80–83.
- [64] \_\_\_\_\_, The variety of moduli of curves of genus four is rational, Dokl. Akad. Nauk SSSR 290 (1986), no. 6, 1292–1294; English transl., Soviet Math. Dokl. 34 (1987), no. 2, 388–389.
- [65] \_\_\_\_\_, Rationality of moduli varieties of plane curves of degree 3k, Mat. Sb. (N.S.)
   136(178) (1988), no. 3, 377–383, 431; English transl., Math. USSR-Sb. 64 (1989), no. 2, 375–381.
- [66] \_\_\_\_\_, Stable rationality of fields of invariants of linear representations of the groups PSL<sub>6</sub> and PSL<sub>12</sub>, Mat. Zametki **48** (1990), no. 2, 49–52, 159; English transl., Math. Notes **48** (1990), no. 1-2, 751–753 (1991).
- [67] \_\_\_\_\_, Rationality of the variety of moduli of curves of genus 5, Mat. Sb. 182 (1991), no. 3, 457–464; English transl., Math. USSR-Sb. 72 (1992), no. 2, 439–445.
- [68] \_\_\_\_\_, Birational geometry of moduli varieties of vector bundles over P<sup>2</sup>, Izv. Akad. Nauk SSSR Ser. Mat. 55 (1991), no. 2, 429–438; English transl., Math. USSR-Izv. 38 (1992), no. 2, 419–428.
- [69] \_\_\_\_\_, On the birational geometry of the space of ternary quartics, Lie groups, their discrete subgroups, and invariant theory, 1992, pp. 95–103.
- [70] \_\_\_\_\_, Rationality of the moduli variety of mathematical instantons with  $c_2 = 5$ , Lie groups, their discrete subgroups, and invariant theory, 1992, pp. 105–111.
- [71] \_\_\_\_\_, Rationality of the moduli variety of curves of genus 3, Comment. Math. Helv.
   71 (1996), no. 4, 507–524.
- [72] M. Kervaire and T. Vust, Fractions rationnelles invariantes par un groupe fini: quelques exemples, Algebraische Transformationsgruppen und Invariantentheorie, 1989, pp. 157– 179.
- [73] I. Kersten, Noether's problem and normalization, Jahresber. Deutsch. Math.-Verein. 100 (1998), no. 1, 3–22.
- [74] M.-A. Knus, A. Merkurjev, M. Rost, and J.-P. Tignol, *The book of involutions*, American Mathematical Society Colloquium Publications, vol. 44, American Mathematical Society, Providence, RI, 1998. With a preface in French by J. Tits.
- [75] J. Kollár, Nonrational hypersurfaces, J. Amer. Math. Soc. 8 (1995), no. 1, 241–249.
- [76] \_\_\_\_\_, Nonrational covers of  $\mathbb{CP}^m \times \mathbb{CP}^n$ , Explicit birational geometry of 3-folds, 2000, pp. 51–71.
- [77] I. Ya. Kolpakov-Miroshnichenko and Yu. G. Prokhorov, Rationality of the field of invariants of the faithful four-dimensional representation of the icosahedral group, Mat. Zametki 41 (1987), no. 4, 479–483, 619; English transl., Math. Notes 41 (1987), no. 3-4, 270–272.
- [78] \_\_\_\_\_, Rationality of fields of invariants of some four-dimensional linear groups, and an equivariant construction connected with the Segre cubic, Mat. Sb. 182 (1991), no. 10, 1430–1445; English transl., Math. USSR-Sb. 74 (1993), no. 1, 169–183.
- [79] \_\_\_\_\_, Construction of the rationality of fields of invariants of some finite fourdimensional linear groups that are connected with Fano threefolds, Mat. Zametki 51 (1992), no. 1, 114–117; English transl., Math. Notes 51 (1992), no. 1-2, 74–76.
- [80] V. É. Kordonskiĭ, Stable rationality of the group Spin<sub>10</sub>, Uspekhi Mat. Nauk 55 (2000), no. 1(331), 171–172; English transl., Russian Math. Surveys 55 (2000), no. 1, 178–179.
- [81] \_\_\_\_\_, Essential dimension and stable rationality of algebraic groups, Vestnik Moskov. Univ. Ser. I Mat. Mekh. (2000), no. 2, 18–21, 64; English transl., Moscow Univ. Math. Bull. 55 (2000), no. 2, 18–21.

- [82] \_\_\_\_\_, On the birational classification of actions of algebraic groups, Izv. Ross. Akad. Nauk Ser. Mat. 65 (2001), no. 1, 61–76; English transl., Izv. Math. 65 (2001), no. 1, 57–70.
- [83] H. Kraft, Geometrische Methoden in der Invariantentheorie, Aspects of Mathematics, D1, Friedr. Vieweg & Sohn, Braunschweig, 1984.
- [84] L. Le Bruyn, Some remarks on rational matrix invariants, J. Algebra 118 (1988), no. 2, 487–493.
- [85] \_\_\_\_\_, Centers of generic division algebras, the rationality problem 1965–1990, Israel J. Math. 76 (1991), no. 1-2, 97–111.
- [86] M. Lorenz, Multiplicative invariant theory, Encyclopaedia Math. Sci., vol. 135, Springer, Berlin, 2005.
- [87] D. Luna, Slices étales, Sur les groupes algébriques, 1973, pp. 81–105. Bull. Soc. Math. France, Paris, Mémoire 33.
- [88] D. Luna and R. W. Richardson, A generalization of the Chevalley restriction theorem, Duke Math. J. 46 (1979), no. 3, 487–496.
- [89] D. Luna and Th. Vust, Un théorème sur les orbites affines des groupes algébriques semi-simples, Ann. Scuola Norm. Sup. Pisa (3) 27 (1973), 527–535 (1974).
- [90] T. Maeda, Noether's problem for  $A_5$ , J. Algebra **125** (1989), no. 2, 418–430.
- [91] J. McCleary, A user's guide to spectral sequences, 2nd ed., Cambridge Studies in Advanced Mathematics, vol. 58, Cambridge University Press, Cambridge, 2001.
- [92] A. S. Merkurjev, Unramified cohomology of classifying varieties for classical simply connected groups, Ann. Sci. École Norm. Sup. (4) 35 (2002), no. 3, 445–476.
- [93] T. Miyata, Invariants of certain groups. I, Nagoya Math. J. 41 (1971), 69–73.
- [94] D. Mumford and J. Fogarty, Geometric invariant theory, 2nd ed., Ergebnisse der Mathematik und ihrer Grenzgebiete, vol. 34, Springer-Verlag, Berlin, 1982.
- [95] M. Nagata, Note on orbit spaces, Osaka Math. J. 14 (1962), 21–31.
- [96] M. Nagata, Lectures on the fourteenth problem of Hilbert, Tata Institute of Fundamental Research, Bombay, 1965.
- [97] P. E. Newstead, Introduction to moduli problems and orbit spaces, Tata Institute of Fundamental Research Lectures on Mathematics and Physics, vol. 51, Tata Institute of Fundamental Research, Bombay, 1978.
- [98] M. Ojanguren, *The Witt group and the problem of Lüroth*, Dottorato di Ricerca in Matematica. [Doctorate in Mathematical Research], ETS Editrice, Pisa, 1990. With an introduction by Inta Bertuccioni.
- [99] E. Peyre, Unramified cohomology and rationality problems, Math. Ann. 296 (1993), no. 2, 247–268.
- [100] \_\_\_\_\_, Unramified cohomology in degree 3 and Noether's problem, available at arXiv: math.AG/0212039.
- [101] A. Pfister, Quadratische Formen in beliebigen Körpern, Invent. Math. 1 (1966), 116– 132.
- [102] B. Plans, On the  $\mathbb{Q}$ -rationality of  $\mathbb{Q}(X_1, \ldots, X_5)^{\mathfrak{A}_5}$  (Preprint, 2005).
- [103] V. L. Popov, Stability criteria for the action of a semisimple group on a factorial variety, Izv. Akad. Nauk SSSR Ser. Mat. 34 (1970), no. 3, 523–531 (Russian); English transl., Math. USSR Izv. 4 (1970), no. 3, 527–535.
- [104] \_\_\_\_\_, Modern developments in invariant theory, Proceedings of the International Congress of Mathematicians, Vol. 1, 2 (Berkeley, Calif., 1986), 1987, pp. 394–406.
- [105] \_\_\_\_\_, Invariant theory, Algebra and analysis (Kemerovo, 1988), 1991, pp. 99–112.
- [106] \_\_\_\_\_, On the "lemma of Seshadri", Lie groups, their discrete subgroups, and invariant theory, 1992, pp. 167–172.
- [107] \_\_\_\_\_, Sections in invariant theory, The Sophus Lie Memorial Conference (Oslo, 1992), 1994, pp. 315–361.
- [108] V. L. Popov and È. B. Vinberg, Invariant theory, Algebraic geometry, 4 (Russian), 1989, pp. 137–314, 315; English transl., Linear algebraic groups, invariant theory, Algebraic geometry. IV, 1994, pp. 123–278.

- [109] C. Procesi, Non-commutative affine rings, Atti Accad. Naz. Lincei Mem. Cl. Sci. Fis. Mat. Natur. Sez. I (8) 8 (1967), 237–255.
- [110] \_\_\_\_\_, Relazioni tra geometria algebrica ed algebra non commutativa. Algebre cicliche e problema di Lüroth, Boll. Un. Mat. Ital. A (5) 18 (1981), no. 1, 1–10.
- [111] Yu. G. Prokhorov, Fields of invariants of finite linear groups (Preprint 2005), available at http://www.math.princeton.edu/~ytschink/papers/.rationality/ rationality-index.html.
- [112] A. V. Pukhlikov, Birational automorphisms of Fano hypersurfaces, Invent. Math. 134 (1998), no. 2, 401–426.
- [113] A. V. Pukhlikov, Birationally rigid Fano hypersurfaces, Izv. Ross. Akad. Nauk Ser. Mat. 66 (2002), no. 6, 159–186; English transl., Izv. Math. 66 (2002), no. 6, 1243– 1269.
- [114] M. Rosenlicht, Some basic theorems on algebraic groups, Amer. J. Math. 78 (1956), 401–443.
- [115] \_\_\_\_\_, A remark on quotient spaces, An. Acad. Brasil. Ci. **35** (1963), 487–489.
- [116] \_\_\_\_\_, Another proof of a theorem on rational cross sections, Pacific J. Math. 20 (1967), 129–133.
- [117] D. J. Saltman, Generic Galois extensions and problems in field theory, Adv. in Math. 43 (1982), no. 3, 250–283.
- [118] \_\_\_\_\_, Retract rational fields and cyclic Galois extensions, Israel J. Math. 47 (1984), no. 2-3, 165–215.
- [119] \_\_\_\_\_, Noether's problem over an algebraically closed field, Invent. Math. 77 (1984), no. 1, 71–84.
- [120] \_\_\_\_\_, Groups acting on fields: Noether's problem, Group actions on rings (Brunswick, Maine, 1984), 1985, pp. 267–277.
- [121] \_\_\_\_\_, The Brauer group and the center of generic matrices, J. Algebra 97 (1985), no. 1, 53–67.
- [122] \_\_\_\_\_, Multiplicative field invariants, J. Algebra 106 (1987), no. 1, 221–238.
- [123] \_\_\_\_\_, Galois groups of order  $p^3$ , Comm. Algebra 15 (1987), no. 7, 1365–1373.
- [124] \_\_\_\_\_, Invariant fields of linear groups and division algebras, Perspectives in ring theory (Antwerp, 1987), 1988, pp. 279–297.
- [125] \_\_\_\_\_, Twisted multiplicative field invariants, Noether's problem, and Galois extensions, J. Algebra 131 (1990), no. 2, 535–558.
- [126] \_\_\_\_\_, Multiplicative field invariants and the Brauer group, J. Algebra 133 (1990), no. 2, 533–544.
- [127] \_\_\_\_\_, A note on generic division algebras, Abelian groups and noncommutative rings, 1992, pp. 385–394.
- [128] \_\_\_\_\_, Brauer groups of invariant fields, geometrically negligible classes, an equivariant Chow group, and unramified H<sup>3</sup>, K-theory and algebraic geometry: connections with quadratic forms and division algebras (Santa Barbara, CA, 1992), 1995, pp. 189– 246.
- [129] \_\_\_\_\_,  $H^3$  and generic matrices, J. Algebra **195** (1997), no. 2, 387–422.
- [130] \_\_\_\_\_, Ramification and lattices, J. Algebra 195 (1997), no. 2, 423–464.
- [131] \_\_\_\_\_, A nonrational field, answering a question of Hajja, Algebra and number theory (Fez), 2000, pp. 263–271.
- [132] \_\_\_\_\_, Invariant fields of symplectic and orthogonal groups, J. Algebra 258 (2002), no. 2, 507–534.
- [133] D. J. Saltman and J.-P. Tignol, Generic algebras with involution of degree 8m, J. Algebra 258 (2002), no. 2, 535–542.
- [134] J.-J. Sansuc, Groupe de Brauer et arithmétique des groupes algébriques linéaires sur un corps de nombres, J. Reine Angew. Math. 327 (1981), 12–80.
- [135] A. Schofield, Matrix invariants of composite size, J. Algebra 147 (1992), no. 2, 345– 349.

- [136] J.-P. Serre, Corps locaux, Hermann, Paris, 1968. Deuxième édition; Publications de l'Université de Nancago, No. VIII.
- [137] \_\_\_\_\_, Cohomologie galoisienne, 5th ed., Lecture Notes in Mathematics, vol. 5, Springer-Verlag, Berlin, 1994.
- [138] C. S. Seshadri, On a theorem of Weitzenböck in invariant theory, J. Math. Kyoto Univ. 1 (1961/1962), 403–409.
- [139] I. R. Shafarevich, *The Lüroth problem*, Trudy Mat. Inst. Steklov. 183 (1990), 199–204, 229 (Russian); English transl., Proc. Steklov Inst. Math. 1991 (1991), no. 4, 241–246.
- [140] N. I. Shepherd-Barron, The rationality of certain spaces associated to trigonal curves, Algebraic geometry, Bowdoin, 1985 (Brunswick, Maine, 1985), 1987, pp. 165–171.
- [141] \_\_\_\_\_, The rationality of some moduli spaces of plane curves, Compositio Math. 67 (1988), no. 1, 51–88.
- [142] \_\_\_\_\_, Invariant theory for  $S_5$  and the rationality of  $M_6$ , Compositio Math. **70** (1989), no. 1, 13–25.
- [143] \_\_\_\_\_, Rationality of moduli spaces via invariant theory, Topological methods in algebraic transformation groups (New Brunswick, NJ, 1988), 1989, pp. 153–164.
- [144] T. A. Springer and R. Steinberg, *Conjugacy classes*, Seminar on Algebraic Groups and Related Finite Groups (The Institute for Advanced Study, Princeton, N.J., 1968/69), 1970, pp. 167–266.
- [145] H. Sumihiro, Equivariant completion, J. Math. Kyoto Univ. 14 (1974), 1–28.
- [146] R. G. Swan, Invariant rational functions and a problem of Steenrod, Invent. Math. 7 (1969), 148–158.
- [147] \_\_\_\_\_, Noether's problem in Galois theory, Emmy Noether in Bryn Mawr (Bryn Mawr, Pa., 1982), 1983, pp. 21–40.
- [148] J. J. Sylvester, On the involution of two matrices of the second order, British Assoc. Report (Southport) (1983), 430–432; reprinted in Collected Mathematical Papers, Vol. 4, Chelsea, 1973.
- [149] B. Totaro, The Chow ring of a classifying space, Algebraic K-theory (Seattle, WA, 1997), 1999, pp. 249–281.
- [150] D. D. Triantaphyllou, Invariants of finite groups acting nonlinearly on rational function fields, J. Pure Appl. Algebra 18 (1980), no. 3, 315–331.
- [151] M. Van den Bergh, The center of the generic division algebra, J. Algebra 127 (1989), no. 1, 106–126.
- [152] È. B. Vinberg, Rationality of the field of invariants of a triangular group, Vestnik Moskov. Univ. Ser. I Mat. Mekh. (1982), no. 2, 23–24, 115; English transl., Moscow Univ. Math. Bull. 37 (1982), no. 2, 27–29.
- [153] V. E. Voskresenskiĭ, Algebraic groups and their birational invariants, Translations of Mathematical Monographs, vol. 179, American Mathematical Society, Providence, RI, 1998. Translated from the Russian manuscript by Boris Kunyavski [Boris È. Kunyavskiĭ].
  - \* Université Paris-Sud, 91405 Orsay, France
  - <sup>†</sup> IMJ, case 7012, Université Paris 7, 2 place Jussieu, 75251 Paris, France